

Citation: Bourne ME, Gloder G, Weldegergis BT, Slingerland M, Ceribelli A, Crauwels S, et al. (2023) Parasitism causes changes in caterpillar odours and associated bacterial communities with consequences for host-location by a hyperparasitoid. PLoS Pathog 19(3): e1011262. https://doi.org/10.1371/journal.ppat.1011262

Editor: Adler R. Dillman, University of California Riverside, UNITED STATES

Received: September 30, 2022

Accepted: March 2, 2023

Published: March 22, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1011262

Copyright: © 2023 Bourne et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The sequences obtained in this study were deposited in the Sequence Read Archive (SRA) at NCBI under RESEARCH ARTICLE

Parasitism causes changes in caterpillar odours and associated bacterial communities with consequences for host-location by a hyperparasitoid

Mitchel E. Bourne^{1•}*, Gabriele Gloder^{2,3•}, Berhane T. Weldegergis¹, Marijn Slingerland¹, Andrea Ceribelli¹, Sam Crauwels^{2,3}, Bart Lievens^{2,3}, Hans Jacquemyn^{3,4}, Marcel Dicke¹, Erik H. Poelman^{1*}

1 Laboratory of Entomology, Wageningen University & Research, Wageningen, The Netherlands, 2 CMPG Laboratory for Process Microbial Ecology and Bioinspirational Management (PME&BIM), Department M2S, KU Leuven, Leuven, Belgium, 3 Leuven Plant Institute (LPI), KU Leuven, Leuven, Belgium, 4 Laboratory of Plant Conservation and Population Biology, Biology Department, KU Leuven, Leuven, Belgium

These authors contributed equally to this work.
* mitchel.bourne@wur.nl (MEB); erik.poelman@wur.nl (EHP)

Abstract

Microorganisms living in and on macroorganisms may produce microbial volatile compounds (mVOCs) that characterise organismal odours. The mVOCs might thereby provide a reliable cue to carnivorous enemies in locating their host or prey. Parasitism by parasitoid wasps might alter the microbiome of their caterpillar host, affecting organismal odours and interactions with insects of higher trophic levels such as hyperparasitoids. Hyperparasitoids parasitise larvae or pupae of parasitoids, which are often concealed or inconspicuous. Odours of parasitised caterpillars aid them to locate their host, but the origin of these odours and its relationship to the caterpillar microbiome are unknown. Here, we analysed the odours and microbiome of the large cabbage white caterpillar Pieris brassicae in relation to parasitism by its endoparasitoid Cotesia glomerata. We identified how bacterial presence in and on the caterpillars is correlated with caterpillar odours and tested the attractiveness of parasitised and unparasitised caterpillars to the hyperparasitoid Baryscapus galactopus. We manipulated the presence of the external microbiome and the transient internal microbiome of caterpillars to identify the microbial origin of odours. We found that parasitism by C. glomerata led to the production of five characteristic volatile products and significantly affected the internal and external microbiome of the caterpillar, which were both found to have a significant correlation with caterpillar odours. The preference of the hyperparasitoid was correlated with the presence of the external microbiome. Likely, the changes in external microbiome and body odour after parasitism were driven by the resident internal microbiome of caterpillars, where the bacterium Wolbachia sp. was only present after parasitism. Microinjection of Wolbachia in unparasitised caterpillars increased hyperparasitoid attraction to the caterpillars compared to untreated caterpillars, while no differences were found compared to parasitised caterpillars. In conclusion, our results indicate that host-parasite

Bioproject PRJNA878850. zOTU count data per sample for the internal and external microbiome of caterpillars (10.6084/m9.figshare.21253455), the corresponding measured VOCs (10.6084/m9. figshare.21253518) and the outcome of choice tests (10.6084/m9.figshare.21253524) and nochoice tests (10.6084/m9.figshare.22133318) are available on Figshare.

Funding: This work was supported by NWO (ALW Open Programme ALWOP.343 to M.D., Spinoza award SPI 87-313 to M.D., and ALW Open Programme ALWOP.368 to E.H.P) and the Flemish Fund for Scientific Research (FWO; G.0961.19N to B.L. and H.J.). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

interactions can affect multi-trophic interactions and hyperparasitoid olfaction through alterations of the microbiome.

Author summary

Bacteria living in and on macroorganisms produce the majority of their odours and thereby play an essential role in species interactions. For example, the hyperparasitoid enemies of parasitic wasps can reliably use odours of parasitised caterpillars to find the parasitic wasp larvae developing inside the caterpillar. We studied whether parasitism of caterpillars caused changes in caterpillar odours and associated internal and external bacterial communities to elucidate whether specific bacteria are responsible for the odours emitted by parasitised caterpillars. We identified which bacteria corresponded with odour production and whether hyperparasitoids could still find parasitised caterpillars that had their bacterial communities experimentally disrupted. We found that parasitised caterpillars have different odour profiles compared to unparasitised caterpillars and that bacteria in and on caterpillars correspond with these changes. The hyperparasitoid was less attracted to parasitised caterpillars with a disrupted skin microbiome. A characteristic bacterium, Wolbachia sp., which was only found inside parasitised caterpillars, may have initiated changes in the external microbiome and effects on odour emission. Furthermore, we show that micro-injection of Wolbachia in unparasitized caterpillars enhanced hyperparasitoid attraction. Hence, our results indicate that parasitoid wasps indirectly reveal themselves to their hyperparasitoid natural enemies by alteration of the bacterial communities of their host.

Introduction

It is widely accepted that higher organisms like plants and animals harbour complex microbial communities, including (but not limited to) bacteria, viruses and fungi [1,2]. Organisms are no longer viewed individually, but as "holobionts", referring to the sum of an organism and its full community of microbial associates [3–5]. The microorganismal part of the holobiont (also called microbiome) is known to play a role in the extended phenotype of macroorganisms and can affect host physiology, phenotype and fitness [2,5,6]. These interactions between microorganisms and their host range from mutualistic to pathogenic and parasitic [7].

Insects are among the most speciose and diverse groups of organisms on earth. They overcame many constraints in adaptation to new niches by engaging in mutualistic interactions with microorganisms [7,8]. Besides affecting the host, the insect microbiome can play an important role in intra-and inter-species interactions, extending to interactions between kingdoms [9]. Microorganisms can affect insect olfactory guided behaviour through the production of microorganism-derived volatile organic compounds (mVOCs) [10]. Many insects use mVOCs to navigate a space and find food sources [11], suitable mating partners [12], prey or hosts [13,14].

A substantial amount of animal-associated odours like pheromones, faeces and body odours are tightly associated with the production of mVOCs [10,15]. Organismal odours can contain precise information due to the strong correlation between organismal characteristics and the microbiome, including information about animal identity, sex and life stage [16–18]. Hence, carnivorous and hematophagous insects can eavesdrop upon mVOCs to locate their

prey or (blood) host [13,16,17]. Their preference is often linked to the presence of specific microorganisms producing mVOCs [14,19,20]. For example, mosquitoes are known to be attracted towards individuals with a higher bacterial abundance and lower diversity [13,21]. Bacteria-derived mVOCs are also known to contribute to the characteristic odours of insect faeces (also referred to as frass) [22], which can be used by natural enemies to locate their host or prey [23–25].

Pathogenic microorganisms that challenge the immune system can affect the host microbiome, resulting in fitness consequences when interfering in essential symbioses [26–28]. Microbiome changes can subsequently affect animal-associated odours through an altered production of mVOCs [15]. Organisms with a parasitic lifestyle are known to harness (pathogenic) microorganisms to overcome host immunity [28,29]. A prominent group of insects with a parasitic lifestyle are parasitoid wasps (also called parasitoids) that lay their eggs in or on the body of their host, which are often larval stages of other insects [30]. Koinobiont parasitoids have evolved sophisticated host manipulations due to their intimate relationship with their host. They are highly specialized and let their hosts continue to feed and grow for a significant time after parasitisation [31]. Upon parasitoid oviposition, microbial symbionts such as bacteria and viruses are injected to manipulate host physiology and immune system to benefit the survival of the parasitoid's offspring [32,33]. This is particularly known for caterpillars where parasitism affects the gut microbiome [34,35], which otherwise consists for a large part of plant- and soil-derived transient bacteria [36,37].

Caterpillars of the small cabbage white Pieris rapae (Lepidoptera: Pieridae) are known to have altered body odours after parasitism by the koinobiont endoparasitoid wasp Cotesia glomerata (Hymenoptera: Braconidae) [38]. This allows the hyperparasitoid Baryscapus galactopus (Hymenoptera: Eulophidae), a natural enemy of C. glomerata, to locate parasitised caterpillars and complete its lifecycle by oviposition in the parasitoid larvae [39,40]. Likewise, the internal microbiome of the closely related large cabbage white Pieris brassicae (Lepidoptera: Pieridae) is strongly modified after parasitism, especially its bacterial community composition [41]. It can be hypothesized that the attractiveness of parasitised caterpillars to hyperparasitoids derives from changes in caterpillar odours through the production of mVOCs, as suggested by its altered microbiome. To date, experiments on attractiveness of odours of parasitised caterpillars to hyperparasitoids could not pinpoint whether the odours are derived from the caterpillar body or products such as frass, and what role external or internal microbiomes play in the odour profile of parasitised caterpillars [38,40]. To investigate the origin of parasitoid-induced attractiveness to hyperparasitoids we studied the effects of parasitism on the organismal odours and the bacterial microbiome of *P. brassicae*, while manipulating the presence of the external microbiome and the transient internal microbiome (including frass).

We report that the organismal odours and bacterial communities of *P. brassicae* are significantly affected by parasitism of *C. glomerata*. We identified five signature volatile organic compounds (VOCs) related to parasitised caterpillars. Linking caterpillar odours and bacterial communities allowed us to identify nine zero-radius operational taxonomic units (zOTUs) correlated with the changes in odour profiles. One of these zOTUs, corresponding to *Wolbachia* sp., was exclusively found inside parasitised caterpillars. Furthermore, we found that the hyperparasitoid *B. galactopus* preferred the odours of parasitised caterpillars with an intact external microbiome over caterpillars with a disrupted external microbiome. Moreover, injection of *Wolbachia* sp. into unparasitised caterpillars resulted in enhanced attraction of hyperparasitoids, shown by a shorter time to first contact with the *Wolbachia*-injected caterpillars compared to mock-injected caterpillars. These results indicate that parasitoid wasps reveal themselves indirectly to their hyperparasitoid natural enemies by modifying the bacterial communities of their host.

Results

Parasitism status and external microbiome disruption significantly impact caterpillar odours

To identify how parasitism by the endoparasitoid *C. glomerata* affects the odours emitted by *P. brassicae* caterpillars, we sampled fifth instar stage (L5) unparasitised (Pb) and parasitised (Cg) *P. brassicae* caterpillars. We characterised how caterpillar odours were altered by parasitism, and investigated the impact of internal and external odour sources. For both unparasitised and parasitised caterpillars three different treatments were analysed. (i) untreated caterpillars, (ii) starved caterpillars (ST), and (iii) starved then washed caterpillars to disrupt the external microbiome (ST+EMD). We included starvation to remove transient internal bacteria and frass, and starvation followed by disruption of the external microbiome (referred to as external microbiome disruption treatment) to additionally modify the absolute and relative abundance of some microorganisms in the external microbiome of caterpillars. Caterpillars with the same treatment and parasitism status were divided into groups of eight individuals and subjected to a dynamic headspace collection. Directly after the headspace collection, the same caterpillars were subjected to the collection of their external and internal microbiomes (Fig 1).

A total of 33 different volatile organic compounds (VOCs) were identified in caterpillar odours, categorised in six different chemical classes (Table 1). This dataset was subjected to orthogonal partial least squares discriminant analysis (OPLS-DA), which yielded a significant model (CV-ANOVA: p < 0.001, $R^2 = 0.344$, $Q^2 = 0.276$), with two predicting and one orthogonal components accounting for 33.70%, 6.81% and 20.50% of the total variance, respectively. Fourteen volatile compounds (Fig 2A, 2B and 2C and Table 1) strongly contributed to differences among treatments according to the OPLS-DA as indicated by VIP (Variable in Projection) scores larger than 1. These were mainly categorised as alcohols (6 compounds) and





https://doi.org/10.1371/journal.ppat.1011262.g001

Table 1. Volatile compounds (tentative identification) detected in the headspace of untreated, starved or starved then external microbiome disrupted unparasitised or *C. glomerata* parasitised caterpillars (*P. brassicae*). Only those compounds have been included that were present in all replicates of at least one treatment. Amounts of individual compounds are given as the average of peak height $/10^4$ (SE). Variable Importance in the Projection (VIP) values for the OPLS-DA are also given. Bold face VIP scores are higher than 1 and are considered as the most influental VOCs for separation of the treatments as shown in Fig 3. Statistical differences among treatments for compounds with VIP score > 1 are indicated with superscript letters based on Kruskal-Wallis tests with Dunn's test for multiple comparisons including a Bonferroni correction. Compounds in green are correlated with the presence/absence of frass and compounds highlighted in yellow are correlated with parasitism status, compound in blue is not correlated to either of these. Abbreviations used: Cg = Untreated parasitised caterpillars; Cg-ST = starved parasitised caterpillars; Cp-ST+EMD = starved then external microbiome disrupted parasitised caterpillars. Pb = Untreated unparasitised caterpillars; Pb-ST = starved unparasitised caterpillars; Pb-ST+EMD = starved then external microbiome disrupted unparasitised caterpillars.

Number	Compound and class	RT (min) ¹	ERI	LRI	Pb (n = 10)	Cg (n = 10)	Pb-ST (n = 9)	Cg-ST (n = 10)	Pb-ST+EMD (n = 10)	Cg-ST+EMD (n = 10)	VIP-SCORE
	ALCOHOLS										
1	1-Methoxy-2-propanol	5.8	647	649	1471.9 (411.9)	833.4 (262.7)	149.0 (84.6)	196.6 (46.8)	674.0 (240.0)	958.9 (333.0)	0.69
2	3-Pentanol	6.5	672	673 ²	549.4 (179.6) ^a	261.0 (94.1) ^{ab}	3.2 (2.1) ^c	19.3 (4.0) ^{bc}	3.0 (1.7) ^c	14.7 (5.9) ^{bc}	1.07
3	3-Methyl-3-buten-1-ol	7.3	700	710	$26.4(6.0)^{a}$	37.5 (9.4) ^a	2.0 (1.1) ^c	17.2 (3.4) ^{ab}	$1.6(0.8)^{c}$	6.9 (2.7) ^{bc}	1.26
4	3-Methyl-1-butanol	7.4	705	706 ⁴	55.6 (10.4) ^a	45.8 (11.5) ^{ab}	8,2 (3.3) ^{bc}	26.8 (7.8) ^{abc}	4.1 (2.4) ^c	51.8 (34.4) ^{abc}	1.02
5	(Z)-3-Hexen-1-ol	10.9	856	856 ⁴	725.0 (173.9) ^a	350.0 (117.9) ^a	2.6 (1.1) ^b	1.4 (0.7) ^b	$0.9 (0.6)^{\mathrm{b}}$	1.0 (0.6) ^b	1.17
6	1-Hexanol	11.2	868	871	59.8 (9.5)	28.7 (8.6)	12,7 (3.1)	16.0 (4.6)	7.7 (2.5)	9.8 (3.1)	0.73
7	(Z)-3-Hepten-1-ol	13.1	929	947	0.1 (0.0) ^c	29.5 (9.0) ^a	$0.9 (0.8)^{bc}$	8.8 (3.0) ^{ab}	$0.1 (0.0)^{c}$	4.6 (2.4) ^{bc}	1.45
8	3-Octanol	15.1	997	991	19.6 (3.0) ^a	34.6 (9.9) ^a	3.4 (1.6) ^b	14.9 (3.9) ^a	1.3 (0.5) ^b	8.5 (3.3) ^{ab}	1.23
	ALDEHYDES										
9	(E)-2-Butenal	5.4	632	632	46.5 (9.9)	27.9 (8.3)	20.2 (4.7)	14.8 (4.0)	10.3 (1.8)	10.8 (1.7)	0.71
10	3-Methylbutanal	5.5	635	637	40.1 (7.0)	20.2 (8.4)	5.6 (1.4)	5.0 (1.1)	3.8 (0.9)	7.2 (3.8)	0.76
11	2-Methylbutanal	5.7	643	641	53.9 (9.9)	30.8 (13.7)	6.4 (2.0)	4.5 (0.9)	4.8 (1.0)	6.0 (3.5)	0.82
12	(Z)-2-Pentenal	8.0	736	727 ⁵	17.4 (3.0) ^a	9.1 (2.6) ^a	$0.1 (0.0)^{\rm b}$	0.3 (0.2) ^b	$0.1 (0.0)^{\rm b}$	$0.4 (0.3)^{\rm b}$	1.18
	AROMATICS										
13	Phenylacetaldehyde	16.6	1053	1053	22.3 (3.0)	15.1 (3.6)	6.7 (1.1)	7.1 (0.8)	7.2 (1.3)	7.7 (1.2)	0.73
14	1-Phenylethanol	17.0	1067	1063	10.4 (2.0) ^a	8.0 (2.7) ^a	$0.1 (0.0)^{b}$	$0.1 (0.0)^{\rm b}$	$0.1 (0.0)^{\rm b}$	$0.2 (0.1)^{\rm b}$	1.19
15	2-Phenylethanol	18.6	1122	1120	43.2 (11.4)	29.6 (8.5)	2.5 (1.0)	3.9 (1.1)	2.4 (0.8)	4.1 (1.4)	0.95
16	Benzothiazole	21.8	1249	1238	174.6 (26.5)	111.1 (27.5)	59.7 (17.5)	71.1 (26.1)	66.4 (9.3)	68.5 (15.7)	0.71
17	(E)-Anethole	23.1	1298	1294	32.0 (6.7)	19.5 (4.7)	33.3 (10.7)	16.5 (5.3)	18.1 (4.7)	22.3 (2.7)	0.48
	KETONES										
18	2,3-Butanedione	4.2	<600	588 ³	704.0 (200.5)	344.0 (87.3)	69.1 (33.1)	87.3 (17.0)	74.5 (33.3)	32.3 (8.3)	0.93
19	3-Pentanone	6.4	667	669 ⁴	75.4 (35.9) ^a	35.7 (11.2) ^a	3.2 (0.9) ^b	9.7 (0.9) ^a	2.1 (0.7) ^b	10.2 (2.1) ^a	1.10
20	3-Hydroxy-2-butanone	6.7	678	681	10.1 (2.4)	8.1 (4.2)	0.9 (0.5)	1.0 (0.5)	0.1 (0.0)	0.7 (0.3)	0.95
21	3-Octanone	14.8	986	984	$0.1 (0.0)^{b}$	2.9 (0.9) ^a	$0.1 (0.0)^{\rm b}$	3.0 (0.5) ^a	$0.1 (0.0)^{\rm b}$	2.7 (0.8) ^a	1.78
	NITROGEN AND/OR SULPI	HUR CON	TAINI	NG CO	MPOUNDS						
22	2-Methyl-propanenitrile	4.8	611	623	14.9 (5.7)	17.6 (8.5)	16.4 (9,2)	14.7 (7.2)	34.6 (20.4)	8.2 (2.7)	0.51
23	Dimethyl disulfide	7.8	725	727	878.4 (258.1)	843.5 (313.9)	61.3 (46.8)	33.0 (9.3)	23.7 (10.6)	20.5 (5.4)	0.97
24	Methylthioacetaldehyde	8.2	748	760 ³	19.2 (5.0) ^a	21.7 (11.0) ^a	0.3 (0.2) ^b	0.1 (0.0) ^b	0.1 (0.0) ^b	0.1 (0.0) ^b	1.17
25	Dimethyl trisulfide	14.6	982	988 ³	356.8 (123.1) ^a	788.9 (564.3) ^a	5.4 (2.7) ^b	1.7 (0.7) ^b	7.0 (5.6) ^b	3.4 (1.0) ^b	1.08
26	4-(methylthio)-Butanenitrile	17.6	1089	1092 ³	431.6 (209.9) ^a	127.7 (58.3) ^a	8.9 (7.5) ^b	1.3 (1.2) ^b	0.1 (0.0) ^b	2.4 (1.6) ^b	1.14

(Continued)

Table 1. (Continued)

Number	Compound and class	RT (min) ¹	ERI	LRI	Pb (n = 10)	Cg (n = 10)	Pb-ST (n = 9)	Cg-ST (n = 10)	Pb-ST+EMD (n = 10)	Cg-ST+EMD (n = 10)	VIP-SCORE
27	Methyl (methylthio) methyl disulfide	19.1	1144	1144	108.9 (39.2) ^a	194.1 (118.2) ^a	1.3 (0.5) ^b	0.3 (0.2) ^b	0.2 (0.1) ^b	0.1 (0.0) ^b	1.21
	TERPENOIDS										
28	p-Menth-3-ene	14.9	992	988	14.2 (3.6)	11.8 (5.5)	10.1 (4.7)	12.1 (4.7)	14.6 (4.2)	13.8 (6.1)	0.49
29	1,8-Cineole	16.4	1044	1042	18.9 (3.2)	11.3 (2.6)	1.8 (1.1)	2.8 (0.4)	1.7 (0.7)	2.8 (0.6)	0.99
30	Dihydromyrcenol	17.2	1073	1074	26.4 (6.1)	14.4 (3.3)	6.4 (1.6)	7.7 (1.3)	5.0 (0.7)	7.5 (1.7)	0.89
31	Isomenthol	20.3	1189	1192	17.9 (1.9)	12.3 (2.5)	8.3 (2.4)	8.3 (1.8)	5.8 (1.1)	7.3 (0.9)	0.73
32	Verbenone	21.2	1225	1223	4.2 (0.5)	3.2 (0.6)	2.1 (0.6)	2.4 (0.5)	1.4 (0.5)	1.9 (0.5)	0.80
33	Bornyl acetate	23.1	1302	1304	15.6 (8.9)	7.6 (2.5)	5.7 (1.2)	10.5 (2.8)	6.0 (1.6)	12.1 (4.2)	0.82

ERI: Experimentally obtained retention indices on a ZB-5MS analytical column

LRI: Retention indices obtained from NIST MS library (https://webbook.nist.gov/), on a column with (5%-Phenyl)-methylpolysiloxane stationary phase or equivalent unless stated otherwise.

¹RT: Retention time (in minutes) of compounds in the chromatographic window.

²LRI: Retention indices [42]

³LRI: Retention indices [38]

⁴LRI: Retention indices [43]

⁵LRI on a 100% polydimethylsiloxane (PDMS) or equivalent stationary phase.

https://doi.org/10.1371/journal.ppat.1011262.t001

nitrogen and/or sulphur containing compounds (4 compounds). Parasitism status significantly affected caterpillar odours (PERMANOVA: $F_{1,53} = 4.594$, p = 0.011), from which the top four VOCs in VIP-score (3-octanone, (*Z*)-3-hepten-1-ol, 3-methyl-3-buten-1-ol and 3-octanol) and 3-pentanone were characteristic for parasitised caterpillars (Table 1 highlighted in yellow and Fig 2A,). A compound was considered characteristic for parasitised caterpillars when it was significantly increased for parasitised caterpillars in at least one pairwise comparison with unparasitised caterpillars undergoing the same treatment (e.g. Cg-ST and Pb-ST).

The included treatments allowed us to address the origin of these compounds. Starvation (ST) and starvation followed by external microbiome disruption (ST+EMD) significantly affected caterpillar body odours (PERMANOVA: $F_{2,53} = 17.292$, p < 0.001), but differences between parasitised and unparasitised caterpillars were still present. Pairwise analyses within starved treatments (Pb-ST vs Cg-ST) and starved then external microbiome disrupted caterpillars with different parasitism statuses (Pb-ST+EMD vs Cg-ST+EMD) highlight the importance of 3-octanone, (*Z*)-3-hepten-1-ol, 3-methyl-3-buten-1-ol, 3-octanol and 3-pentanone in characterising parasitism status of caterpillars (S1 and S2 Figs and Table A in S1 Supporting Information). This indicates that the source of these VOCs was elsewhere than the frass and skin of the caterpillar, and that they are likely formed internally in the caterpillar.

Overall, starvation followed by external microbiome disruption (ST+EMD) did not impact caterpillar odours strongly, compared to only starvation (ST). Only 3-methyl-3-buten-1-ol was found to be significantly reduced when comparing starved and starved then external microbiome disrupted treatments of the same parasitism status. Accordingly, the separation in OPLS-DA was less evident between starved and starved then external microbiome disrupted caterpillars with the same parasitism status (Figs 3A, S3 and S4). Untreated caterpillars produced several compounds at a higher level (Figs 2B and 3B), including most nitrogen- and/or sulphur-containing compounds. This is likely due to the presence of frass during the head-space collection of untreated caterpillars. Whether hyperparasitoid attraction to caterpillar odours could be explained by frass was investigated in a two-chamber olfactometer setup as



Fig 2. Volatile compounds (tentative identification) detected in the headspace of untreated, starved or starved then external microbiome disrupted unparasitised or *C. glomerata* parasitised caterpillars (*P. brassicae*) with a Variable Importance in the Projection (VIP) values > 1 for the OPLS-DA. Amounts of individual compounds are given as the average of peak height $/10^4$ (±SE). Statistical differences among treatments for compounds with VIP score > 1 are indicated with different letters based on Kruskal-Wallis tests with Dunn's test for multiple comparisons including a Bonferroni correction. a) Compounds in yellow background are correlated with parasitism status. b) Compounds in green background are correlated with the presence/absence of frass. c) Compound in blue background is not correlated to either of these. Abbreviations used: Pb = Untreated unparasitised caterpillars; Cg = Untreated parasitised caterpillars; Pb-ST + EMD = starved then external microbiome disrupted unparasitised caterpillars. Cg-ST+EMD = starved then external microbiome disrupted parasitised caterpillars.

https://doi.org/10.1371/journal.ppat.1011262.g002



Fig 3. Overview of caterpillar volatile profiles of untreated, starved or starved then external microbiome disrupted *C. glomerata* parasitised or **unparasitised caterpillars** (*P. brassicae*). a) OPLS-DA (Orthogonal Projection to Latent Structures Discriminant Analysis) plot for the volatile blends of different groups of caterpillars. The Hotelling's T2 ellipse confines the confidence region (95%) of the score plot. b) Loading plot defining the contribution of each of the volatile compounds to the separation of treatment groups. Volatile compounds closer to a treatment in the plot correlate stronger with the treatment. For compound identity see <u>Table 1</u>. Abbreviations used: Cg = Untreated parasitised caterpillars; Cg-ST = starved parasitised caterpillar; Cg-ST +EMD = starved then external microbiome disrupted parasitised caterpillars. Pb = Untreated unparasitised caterpillars; Pb-ST = starved unparasitised caterpillars; Pb-ST = starved then external microbiome disrupted unparasitised caterpillars.

https://doi.org/10.1371/journal.ppat.1011262.g003

described in [38]. Our results indicate that frass was attractive over an empty chamber (Empty vs Pb-frass: p = 0.007, U = 643.5; Empty vs Cg-frass: p = 0.163, U = 565), but reject the hypothesis that the hyperparasitoid can use frass to discriminate between parasitised and unparasitised caterpillars (Pb-frass vs Cg-frass: p = 0.504, U = 1066.5) (S5 Fig).

Parasitism status and external microbiome disruption significantly impact caterpillar external and internal microbiome

Because organismal odours are tightly associated with the microorganisms present in or on the body of macroorganisms, we subsequently characterised how the caterpillar microbiome was altered by parasitism, starvation, and starvation followed by external microbiome disruption. Directly after the collection of caterpillar odours, the same caterpillars were subjected to the collection of their external and internal microbiomes (Fig 1). Analysis of the bacterial 16S ribosomal RNA (rRNA) gene sequences revealed a significant effect of parasitism on the external microbiome (PERMANOVA, $F_{1,53} = 5.759$, p = 0.002) of *P. brassicae* caterpillars (Fig 4A). The external microbiome of all caterpillars mainly consisted of three zOTUs (combined >69% relative abundance); *Enterococcus* sp. (zOTU 1), Enterobacteriaceae (zOTU 2) and *Staphylococcus* sp. (zOTU 3) (Fig 4C and Table B in S1 Supporting Information). *Staphylococcus* sp. (zOTU 3) was more abundant on unparasitised caterpillars, while on parasitised caterpillars an



c) External microbiome

Blasted ID	zOTU ID	Pb	Cg	Pb-ST	Cg-ST	Pb-ST+EMD	Cg-ST+EMD	Overall
Enterococcus	zOTU1	32.5	39.7	23.1	48.2	58.9	72.4	45.8
Enterobacteriaceae	zOTU2	7.4	27.7	10.3	32.9	25.5	26.5	21.7
Staphylococcus	zOTU3	29.1	10.0	27.8	9.9	3.5	0.0	13.4
Jeotgalicoccus	zOTU7	6.3	4.0	4.0	2.8	0.5	0.0	2.9
Acinetobacter	zOTU5	1.6	4.3	7.6	1.5	0.9	0.0	2.7
Staphylococcus	zOTU11	7.4	1.4	5.2	0.4	1.3	0.0	2.6
Corynebacterium	zOTU10	4.5	3.6	3.4	1.9	0.2	0.0	2.3
Acinetobacter	zOTU6	2.6	2.2	4.0	0.1	1.2	0.0	1.7
Glutamicibacter	zOTU12	3.0	2.6	2.3	1.4	1.3	0.0	1.7

d) Internal microbiome

Prevalence (%) 100 80 60 40 20 0

Blasted ID	zOTU ID	Pb	Cg	Pb-ST	Cg-ST	Pb-ST+EMD	Cg-ST+EMD	Overall
Enterococcus	zOTU1	85.1	79.9	72.5	84.3	86.9	85.8	82.4
Enterobacteriaceae	zOTU2	11.5	11.0	17.6	8.9	9.7	7.6	11.1
Wolbachia*	zOTU4	0.0	8.2	0.0	6.5	0.0	6.3	3.5
Sphingomonas	zOTU9	0.2	0.3	8.9	0.1	2.1	0.0	2.0

Fig 4. Overview of caterpillar-associated bacterial communities of untreated, starved or starved then external microbiome disrupted *C. glomerata* parasitised or unparasitised caterpillars (*P. brassicae*). a) Non-metric multidimensional scaling (NMDS) ordination plots based on Bray–Curtis distances of Hellinger-transformed relative abundance data of the external bacterial communities. b) NMDS ordination for the internal bacterial communities. c) External bacterial community profiles of the different caterpillar samples. d) Internal bacterial community profiles of the different caterpillars samples. For each zOTU, the average relative abundance (%) for each group is given in the cell as a percentage, whereas the colour indicates prevalence (white is absent). Only bacteria with an overall relative abundance >1.5% are shown in (c) and (d). The full overview is provided as supporting information (Table B in S1 Supporting Information). zOTUs are identified by a BLAST search against type materials in GenBank. When no significant similarity was found with type materials, the BLAST analysis was performed against entire GenBank (indicated with an asterisk). Identifications were performed at genus level; when identical scores were obtained for different genera, identifications were performed at family level. Abbreviations used: Cg = untreated parasitised caterpillars; Cg-ST = starved parasitised caterpillars; Pb-ST = starved unparasitised caterpillars; Pb-ST

https://doi.org/10.1371/journal.ppat.1011262.g004

unidentified member of Enterobacteriaceae (zOTU 2) had a higher relative abundance and the relative abundance of *Staphylococcus* sp. (zOTU 3) was decreased. Moreover, parasitism had a significant effect on the internal (PERMANOVA, $F_{1,53} = 4.865$, p = 0.014) bacterial community supporting a previous study [41] (Fig 4B). The internal microbiome was dominated by *Enterococcus* sp. (zOTU 1) and Enterobacteriaceae (zOTU 2) (combined >90% of relative abundance), but *Staphylococcus* sp. (zOTU 3) was absent. *Wolbachia* sp. (zOTU 4) was only found in parasitised caterpillars (Fig 4D and Table B in S1 Supporting Information).

To further elucidate where the differences in microbiome came from, we starved caterpillars to exclude transient and frass-related microorganisms and, starved then external microbiome disrupted caterpillars to affect microorganisms on the skin as well. Treatment had a significant effect on the external bacterial community (PERMANOVA, $F_{2,53}$ = 5.716, p <0.001). A strong cluster of bacteria collected from parasitised, starved then external microbiome disrupted caterpillars (Cg-ST+EMD) is clearly visible in the NMDS ordination (Fig 4A). Staphylococcus sp. (zOTU 3) almost fully declined after starvation followed by external microbiome disruption and the relative abundance of Enterococcus sp. (zOTU 1) increased. The internal bacterial communities were not substantially affected by treatment (PERMA-NOVA, $F_{2,53} = 1.083$, p = 0.341), indicating that the difference in microbiome between unparasitised and parasitised caterpillars is caused by resident bacteria. Simply put, the parasitism signal in the microbiome was captured in the resident internal microbiome, where Wolbachia sp. (zOTU 4) was only present in parasitised caterpillars. On the external microbiome there was a remarkable decrease in the relative abundance of Staphylococcus sp. (zOTU 3) after parasitism. The Shannon diversity indices of the external and internal microbiomes were significantly affected by parasitism (external: $F_{1,53} = 5.584$, p = 0.022; internal: $F_{1,53} = 4.379$, p = 0.041; ANOVA). The diversity of the external bacterial community decreased after parasitism and the diversity of the internal bacterial community increased. Treatment of caterpillars reduced the diversity of the external, but not the internal microbiome (external: $F_{2,53} = 12.661$, P < 0.001; internal: $F_{2,53} = 2.394$, p = 0.062; ANOVA) (S7 Fig).

Linking microbial communities to caterpillar odours

To study the relationship between caterpillar odours and bacteria, we decided to subset our data by omitting the untreated samples. The presence of frass in these samples was shown to strongly impact the odours, but had a limited effect on the signature odours for parasitism (Table 1), the internal microbiome (Fig 4B) and hyperparasitoid preference in a two-chamber olfactometer setup (S5 Fig). The relationship between the caterpillar microbiome and odours was found to be significant for both the external (ANOVA-like permutation test: $F_{5,33} = 2.704$, $R^2 = 0.29$, p = 0.001) and internal bacterial communities (ANOVA-like permutation test: $F_{4,34} = 2.99$, $R^2 = 0.26$, p = 0.001). A total of nine zOTUs were selected, which contributed most to the variation in odour composition between samples. These zOTUs represented five zOTUs from the external microbiome and four from the internal microbiome, indicated by the vectors in the RDA tri-plots (Fig 5A and 5B). There was no overlap in the selected zOTUs for the internal- and external bacterial communities.

From the selected bacteria located in the internal microbiome *Wolbachia* sp. (zOTU 4) was the most abundant and was only present in parasitised caterpillars. *Sphingomonas* sp. (zOTU 9) had a higher relative abundance in treated unparasitised caterpillars, and was nearly absent in parasitised caterpillars. A so far unidentified bacterium (zOTU 94) and *Proteus* sp. (zOTU 220) had a low relative abundance ($\leq 0.1\%$) and thus likely had a negligible effect on the caterpillar odours (Fig 5D). Accordingly, all selected zOTUs for the external microbiome had a low ($\leq 0.2\%$) overall relative abundance. From these, *Chryseobacterium* sp. (zOTU 49),



c) Selected zOTUs external microbiome

Blasted ID	zOTU ID	Pb-ST	Cg-ST	Pb-ST+EMD	Cg-ST+EMD	Overall
Bradyrhizobium	zOTU18	0.3	0.2	0.0	0.4	0.2
Chryseobacterium	zOTU49	0.2	0.0	0.0	0.0	0.0
Pseudomonas	zOTU53	0.2	0.0	0.1	0.0	0.0
Enterococcus [98.7%]	zOTU60	0.0	0.0	0.0	0.0	0.0
Paenalcaligenes	zOTU72	0.0	0.0	0.4	0.0	0.1

d) Selected zOTUs internal microbiome

Prevalence % 1 1 1 100 80 60 40 20 0

Blasted ID	zOTU ID	Pb-ST	Cg-ST	Pb-ST+EMD	Cg-ST+EMD	Overall
Wolbachia*	zOTU4	0.0	6.5	0.0	6.3	3.5
Sphingomonas	zOTU9	8.9	0.1	2.1	0.0	2.0
Unidentified bacterium* [79.8%]	zOTU94	0.0	0.0	0.1	0.0	0.0
Proteus	zOTU220	0.0	0.0	0.0	0.0	0.0

Fig 5. Forward selection redundancy analyses (RDA) on volatiles of untreated, starved or starved then external microbiome disrupted *C. glomerata* **parasitised or unparasitised caterpillars (***P. brassicae***) and their bacterial communities.** The triplots indicate the volatile composition of the caterpillars (dots), the volatile compounds (numbers as in <u>Table 1</u>) and forward selected zOTUs after 9999 permuations (vectors) for the external (a) and internal (b) bacterial communities. Further, relative abundance-prevalence data of the selected zOTUs are presented for the external (c) and internal (d) bacterial community. The average relative (%) abundance for each group is given in the cell as a percentage, whereas the colour indicates prevalence (white is absent). zOTUs were identified by a BLAST search against type materials in GenBank. When no significant similarity was found with type materials, the BLAST analysis was performed against entire GenBank (indicated with an asterisk). Identifications were performed at genus level. When identity percentages were lower than 99%, the percentage of sequence identity with the GenBank entry is given between brackets. Abbreviations used: Cg-ST = starved parasitised caterpillar; Cg-ST+EMD = starved then external microbiome disrupted parasitised caterpillars; Pb-ST = starved unparasitised caterpillars; Pb-ST+EMD = starved then external microbiome disrupted unparasitised caterpillars.

https://doi.org/10.1371/journal.ppat.1011262.g005

Enterococcus sp. (zOTU 53) and *Paenalcaligenes* sp. (zOTU 72) were exclusively found on unparasitised caterpillars, whereas *Bradyrhizobium* sp. (zOTU 18) had an increased relative abundance on parasitised caterpillars (Fig 5C). Interestingly, Enterobacteriaceae sp. (zOTU 2) and *Staphylococcus* sp. (zOTU 3) were highly abundant in the external microbiome and affected by parasitism in their relative abundance, but were not selected in the forward selection RDA (Fig 4C). These results point towards a microbe-derived signal of parasitism that comes from within the caterpillar and determines caterpillar odours, while external bacterial communities and frass are less likely to play a role in the production of volatile compounds associated with parasitism status.

Hyperparasitoid females are no longer attracted to external microbiome disrupted parasitised caterpillars

A total of 1000 *B. galactopus* females were tested in two-choice assays in a Y-tube olfactometer for their preference to caterpillar odours. Hyperparasitoids did respond to caterpillar odours and 84.4% of the tested females made a choice within 10 min (Fig 6). The hyperparasitoid preferred the odours of starved, parasitised *P. brassicae* (Cg-ST) over an empty arm (binomial test, p < 0.001) and over odours from caterpillars that had received the external microbiome disruption treatment after starvation (Cg-ST+EMD) (binomial test, p < 0.001) (Fig 6). Disruption of the external microbiome caused a loss of preference for caterpillar odours. Hyperparasitoid preference did not differ between differently treated unparasitised caterpillars (Pb-ST +EMD vs Pb-ST; binomial test, p = 0.28) or starved then external microbiome disrupted caterpillars with different parasitism statuses (Pb-ST+EMD vs Cg-ST+EMD; binomial test, p = 0.06). When the odours of unparasitised versus parasitised starved caterpillars were offered, no preference was found (Pb-ST vs Cg-ST; binomial test, p = 0.76) contrasting the



Fig 6. Preference of the hyperparasitoid *B. galactopus* for caterpillar body odours of *C. glomerata* parasitised and unparasitised *P. brassicae* caterpillars. This was tested with Y-tube olfactometer tests. Tested combinations were selected according to the hypothesis that hyperparasitoids can use caterpillar body odours, which are (at least partially) determined by the external microbiome. Numbers between brackets indicate the number of wasps that made a choice within 10 min from the start of the experiment versus the total number of wasps tested. *** P < 0.001; two-sided binomial test. Abbreviations used: Air = clean air (black); Cg-ST = starved parasitised caterpillar (solid blue); Cg-ST+EMD = starved and external microbiome disrupted parasitised caterpillars (blue border); Pb-ST+EMD = starved then external microbiome disrupted unparasitised caterpillars (red border).

https://doi.org/10.1371/journal.ppat.1011262.g006

comparisons of their microbiome, odour profiles (S3 and S4 Figs) and earlier findings in a closely related caterpillar species (*P. rapae*) [38]. Highly similar results were found when we repeated this experiment using the same species of caterpillar and hyperparasitoid, but parasitised the caterpillars with *Hyposoter ebeninus*, which is another parasitoid host species of the hyperparasitoid (S6 Fig). The consistency of these results indicate that the preference of the hyperparasitoid is correlated with the presence of the external microbiome, though it is not discriminating between the odours of parasitised and unparasitised caterpillars in this setup.

Direct evidence for the role of Wolbachia in hyperparasitoid host location

In order to assess whether the presence of Wolbachia affects hyperparasitoid behaviour, a nochoice experiment was performed in which B. galactopus individuals were subjected to P. brassicae caterpillars injected with Wolbachia sp. (Wolb-PBS, injected once with Wolbachia; Wolb-Wolb, injected twice with Wolbachia) in comparison with caterpillars parasitised by C. glomerata (Parasitised-PBS) or unparasitised caterpillars (Unparasitised-PBS) (Fig 7). Within one hour of release, B. galactopus responded by mounting caterpillars in 84.7% of the cases, and the lowest response rate was observed for unparasitised individuals (75.0%). Significant differences in the time used to make first contact and mount the caterpillars were found between treatments (Kruskal-Wallis $\chi^2 = 11.24$, p = 0.010). Pairwise comparisons showed significant differences in time to first contact between unparasitised and parasitised caterpillars (Dunn's test Z = 2.95, p = 0.019) and between unparasitised caterpillars and those injected once with *Wolbachia* (Z = 2.62, p = 0.043). On average, the wasps needed 21 min 02 sec before mounting unparasitised caterpillars, while it took less time, to mount parasitised caterpillars (10 min 27 sec), caterpillars injected once with Wolbachia (12 min 16 sec) and caterpillars injected two times with Wolbachia (16 min 36 sec), respectively (Fig 7A). Differences between treatments were also statistically significant for the time spent mounting (Kruskal-Wallis χ^2 = 10.241, p = 0.017). Pairwise comparisons revealed statistically significant differences in total mounting time between unparasitised and parasitised caterpillars (Z = -2.85, p = 0.026), and between parasitised caterpillars and those injected once with Wolbachia (Z = 2.77, p = 0.027). On average, hyperparasitoids spent 57 min 55 sec on parasitised individuals, while they spent 50 min 50 sec on healthy caterpillars, 50 min 22 sec on caterpillars injected once with Wolbachia and 53 min 52 sec on caterpillars injected twice with Wolbachia (Fig 7B).

Discussion

In this study we show that parasitism by *C. glomerata* leads to a significant alteration of the odours of its host caterpillar *P. brassicae*, in which five VOCs are considered to be induced after parasitism. This aligns with changes in the internal and external bacterial communities, hinting at a relationship between caterpillars' body odours and their microbiome. Moreover, hyperparasitoids preferred the odours of starved, parasitised caterpillars (Cg-ST) over odours from starved then external microbiome disrupted parasitised caterpillars (Cg-ST+EMD), indicating a role of bacteria and/or their volatiles in the host-location of hyperparasitoids. The role of the internal microbiome was confirmed by the enhanced attraction of the hyperparasitoid to unparasitised caterpillars.

Analysis of the caterpillar headspace composition identified some consistent patterns in alterations of parasitised caterpillar odours regardless of whether caterpillars were starved (ST) or starved then external microbiome disrupted (ST+EMD). This indicates that the external microbiome and transient internal microbiome do not play a direct role in the production of characteristic VOCs. We found a significant association between parasitism and the caterpillar microbiome in the same sets of caterpillars. Both internal and external bacterial communities





https://doi.org/10.1371/journal.ppat.1011262.g007

had a significant correlation with the caterpillar odour profile and we found nine bacterial zOTUs to be correlated with the VOCs produced by caterpillars. These bacteria were especially abundant in the internal microbiome, where the intracellular, resident insect symbiont *Wolbachia* sp. was only found in parasitised caterpillars.

Experimental injection of *Wolbachia* into unparasitised *P. brassicae* caterpillars had a significant effect on hyperparasitoid behaviour. Specifically, hyperparasitoids more rapidly made first contact with *Wolbachia*-injected caterpillars, unparasitized caterpillars compared to mock-injected unparasitized caterpillars. This response to *Wolbachia*-injected caterpillars was similar to parasitised caterpillars. In contrast, total mounting time was not affected by *Wolbachia*-injection, suggesting that this bacterium plays an important role in attracting the hyperparasitoids, but does not arrest the hyperparasitoid on the caterpillar. Most likely, other cues linked to the presence of parasitoid larvae, like mechanosensory cues, are necessary for hyperparasitoids to identify host presence in caterpillars [38].

The VOCs (Z)-3-hepten-1-ol, 3-octanol, 3-pentanone, 3-methyl-3-buten-1-ol and 3-octanone were characteristic for parasitised P. brassicae caterpillars in the L5 stage and can potentially be used by its hyperparasitoids to reliably locate their host. Most of these characteristic compounds are known to be produced as (alarm) pheromones by other insects [44-47], but little is known about (Z)-3-hepten-1-ol. A parasitic fly is known to use an ant alarm pheromone, consisting of 3-octanone, 3-octanol and 3-nonanone, to locate its host [45]. 3-Methyl-3-buten-1-ol is an olfactory-active compound to a coccinellid predator to locate and perceive its aphid prey [48]. Additionally, a blend of 3-octanone and 3-octanol is released upon death of a social termite, allowing conspecifics to recycle the nutrients from the corpse before decomposition starts with higher chances of pathogen infection [46]. These compounds may also be a signal of impending death for parasitised *P. brassicae*, which benefits conspecifics in avoiding natural enemies but also provides its hyperparasitoid with a reliable cue to locate its host. Interestingly, none of the characteristic VOCs for parasitised caterpillars were detected in an earlier study on body odours of the closely related P. rapae, parasitised by the same parasitoid species C. glomerata [38]. It is therefore less likely that characteristic compounds are directly produced by the parasitoid larvae growing inside the caterpillar. In P. rapae, 2,3-butanedione was the only VOC found at higher levels in parasitised caterpillars [38], but it had a low impact on the separation between treatments in our current study (VIP score < 1). This indicates that the alteration of organismal odours after parasitism is species-specific and not overruled by parasitism, as known for plant odours induced by parasitised caterpillars [49]. Nearly all VOCs identified here as frass-related were previously found in caterpillar frass of both P. rapae and P. brassicae (excluding only 1-phenylethanol) [23], confirming their origin and indicating that frass odour is more dependent on food plant than caterpillar species as both caterpillars were reared on Brussels sprouts plants.

Our data show that the internal and external microbiome are altered after parasitism and have a significant correlation with the caterpillar odour profile. This is in line with our expectation that alterations in the microbiome coincide with changes in caterpillar odours through the production of mVOCs. Among their presence in various taxa, four out of five characteristic VOCs for parasitised caterpillars are known to be mVOCs according to the mVOC 3.0 database [50]. 3-Pentanone, 3-methyl-3-buten-1-ol and 3-octanone are produced by a variety of bacteria and fungi, whereas 3-octanol is only reported to be produced by fungi [50–52]. We did not find evidence that these compounds are directly produced by the selected bacteria in this study. Furthermore, it cannot be excluded that fungi play a role in the production of caterpillar odours, but we did not study fungal communities, as they occur in low abundance in association to *P. brassicae* [41]. To improve our understanding of the relationship between selected bacteria, their produced mVOCs and trophic interactions, future studies should consider to perform fermentations with these bacteria, followed by collection and characterisation of their headspace. This can be combined with behavioural assays of (hyper)parasitoids to evaluate their effect on trophic interactions [14,20].

Our results indicate that the internal bacterial communities were not substantially affected by starvation and starvation followed by external microbiome disruption treatments. Thus resident internal bacteria are most likely to be responsible for the difference between parasitised and unparasitised caterpillars. Due to our sampling method these bacteria can originate from the gut, haemolymph or any other internal tissues, including parasitoid larvae present in parasitised caterpillars. In accordance with previous studies, we found that the internal microbiome of *P. brassicae* mainly consisted of *Enterococcus* sp. and Enterobacteriaceae [41,53,54]. Among selected bacteria which contribute most to the variation in odour composition *Wolbachia* sp. and *Sphingomonas* sp. were most abundant and both are present in the resident internal microbiome. *Sphingomonas* sp. was especially abundant in unparasitised caterpillars and *Wolbachia* sp. was only present after parasitism confirming previous data [41]. *Sphingomonas* is known to be a pathogen-repressing- and plant-growth-promoting rhizobacterium. Its genome indicates that it has properties to produce terpenoids and degrade aromatic compounds [55]. However, both terpenoids and aromatic compounds are not involved in differences between unparasitised and parasitised caterpillars. The performed amplicon sequencing does not allow us to explore the genome of the *Sphingomonas* bacterium detected into more depth. Isolation and culturing of this bacterium and performing fermentations and characterisation of the produced mVOCs are needed to further investigate its biosynthetic capacities.

The exclusive presence of *Wolbachia* in parasitised caterpillars can be explained by the fact that it is transferred upon oviposition from the parasitoid into the caterpillar. Moreover, it has been shown in high relative abundance in parasitoid larvae [29,41]. *Wolbachia* is known to be involved in susceptibility to (hyper)parasitoids [56,57], but the mechanism behind this and its relationship to mVOCs and organismal odours are still unknown [56]. Our experiments involving micro-injections with *Wolbachia* confirm the hypothesis that *Wolbachia* is involved in hyperparasitoid host location, but further research is needed to fully understand its contribution.

Wolbachia is mostly known for its ability to alter the reproductive system, behaviour and metabolism of its insect host [58]. Moreover, it has been demonstrated that *Wolbachia*'s presence influences cuticular hydrocarbons and odour-mediated mate preference in a terrestrial isopod [59]. It can be hypothesised that the presence of *Wolbachia* in the internal microbiome after parasitism alters the caterpillar's odour through various mechanisms, which in turn may lead to changes in hyperparasitoid behaviour. For example, changes in the caterpillar's odour profile could be caused (i) by directly producing mVOCs which diffuse to the outside of caterpillars through the spiracles, (ii) by impacting caterpillar metabolism, thus providing other bacteria with a different set of metabolites, which are fermented to signature volatile compounds, or (iii) by indirectly affecting the bacterial communities, and thus the production of mVOCs through its presence. Its obligate intracellular lifestyle makes it difficult to cultivate the bacterium and carry out fermentations or other experiments [14,20]. However, insect cell-lines could provide a solution to characterise *Wolbachia*'s volatile metabolites [60].

Behavioural assays indicate that not frass odours, but the presence of the external microbiome on caterpillars are associated with the preference of *B. galactopus* (Figs 6 and S6). The external caterpillar microbiome is affected by parasitism, but the selected zOTUs on the external microbiome are low in relative abundance, making them poor candidates to produce a substantial amount of mVOCs. This is confirmed by negligible differences in body odours between starved (ST) and, starved and external microbiome disrupted caterpillars (ST+W +EMD) (S3 and S4 Figs). The diversity of the skin microbiome and presence of individual bacteria can also play a role in attracting insects [13,14,16,19,20,61]. Especially parasitised caterpillars undergoing the external microbiome disruption treatment after starvation had a lower diversity of bacteria on their skin, which could explain their reduced attractiveness to the hyperparasitoid (S7 Fig).

The hyperparasitoid *B. galactopus* was unable to discriminate between starved parasitised and starved unparasitised *P. brassicae* caterpillars based on their odours, as opposed to previous findings in *P. rapae*. Possibly, the starvation treatment applied in our current study with *P. brassicae* reduced the differences between parasitised and unparasitised caterpillars on which hyperparasitoids rely. These manipulations were not performed in the previous study on *P. rapae* [38]. The compound 2,3-butadione could play an important role in the ability of hyperparasitoids to discriminate between healthy and parasitised caterpillars as shown for *P. rapae* [38]. The presence of this compound is not strongly affected by parasitism in *P. brassicae*, making it more difficult to differentiate between parasitised and unparasitised *P. brassicae* for this hyperparasitoid when relying on this cue. We cannot rule out that in addition to volatiles, also contact cues are used to determine parasitism status by the hyperparasitoid. Cuticular hydrocarbons are known to be important for aphid hyperparasitoids to locate their host [62].

We found that both the external and internal microbiome of *P. brassicae* caterpillars are altered after parasitism by *C. glomerata.* In contrast, our prior study indicated that *C. glomerata* parasitism only affected the internal microbiome of *P. brassicae* [47]. Furthermore, in our previous study we found a reduction in the diversity of bacterial communities for both internal and external microbiomes after parasitism, whereas our current data identified relatively weak effects. Both studies are on the same study system, but previous findings [7,63–65] suggest that habitat is one of the strongest factors influencing the external microbiome of insects, outweighing the effects of parasitism due to differences in day, weather, environment and/or host plant [41]. As a result, parasitism status played an insignificant role in explaining variation in the external microbiome. To further investigate differences caused by parasitism, our current study pooled the microbiomes of groups of caterpillars and only examined one habitat (lab strain). This allowed us to identify that the external microbiome is altered alongside the internal microbiome. The difference in responses of Shannon diversity can most likely be attributed to the studied lab strain with an overall less diverse bacterial community [41] and to the higher resolution with more statistical power in this study.

In conclusion, our findings provide evidence that parasitism-induced changes in the caterpillar microbiome are responsible for alterations of its odours. For hyperparasitoids, the presence of the skin microbiome is more important than frass odours to locate its host. This is likely mediated by differences in diversity of the external microbiome as well as differentially abundant bacteria between parasitised and unparasitised caterpillars. We identified five VOCs related to parasitism, but did not confirm causality between specific bacteria and these caterpillar odours. The intracellular bacterium Wolbachia sp., which was only found in the internal microbiome of parasitised caterpillars, is the strongest candidate to cause the production of characteristic mVOCs for parasitised caterpillars although the mechanisms are still unclear. It may produce mVOCs directly by itself, but can also affect the bacterial community or the presence of metabolites. Micro-injections of caterpillars with Wolbachia confirmed that Wolbachia is involved in attraction of hyperparasitoids, but could not fully explain their behaviour. Further research is needed to elucidate the role of this intracellular bacterium and its products in the interaction between the host microbiome and its odours, which can be exploited by hyperparasitoids to locate their host. The identified compounds related to parasitised caterpillars can potentially be employed in optimizing biological control that is hampered by hyperparasitoids reducing the population of parasitoids that are used in biological control of herbivorous pests [66]. The bacterial odours (or synthetic counterparts) may be used in traps or a push-pull system to reduce hyperparasitoid populations in cropping systems [67]. However, this would first require a further evaluation of the attractiveness of these compounds on hyperparasitoids and non-target organisms.

Materials and methods

Insects and rearing

Used insects. The larval stage (caterpillar) of the large cabbage white, *P. brassicae*, was selected as focal study object. Its parasitoid *C. glomerata* was included because it is known to

influence the caterpillars' microbiome after parasitism [41]. To elucidate the role of caterpillar odours in the host-location of higher trophic levels we used the hyperparasitoid *B. galactopus*. This hyperparasitoid is known to use caterpillar body odours during host location [38]. *Pieris brassicae* is an important oligophagous pest species whose larvae feed on many members of the Brassicaceae family such as cabbage, cauliflower, Brussels sprouts and rapeseed. In nature, *Pieris* spp. are attacked by various natural enemies, including highly specialised parasitoid wasps, known to have host-manipulating traits [29]. *Cotesia glomerata* is a gregarious koinobiont parasitoid, which parasitises a wide range of *Pieris* spp., with *P. brassicae* and *P. rapae* as its main hosts. The hyperparasitoid *B. galactopus* is widely distributed all around Europe and has been reported as one of the most common hyperparasitoids in the *Brassica oleracea*-associated insect community with a wide host range that includes braconid and ichneumonid parasitoids [38,68]. It is a true hyperparasitoid that parasitises the larvae of parasitoids that are concealed in the herbivore body [40].

Insect rearing conditions. Experiments were performed using lab cultures of *P. brassicae*, C. glomerata and B. galactopus, which were originally collected from agricultural fields in the near vicinity of Wageningen University & Research, the Netherlands. Pieris brassicae was reared in cages and maintained on Brussels sprouts (B. oleracea L. var. gemmifera cv. Cyrus) in greenhouse conditions (± 21°C, 25-35% RH, 16:8 h L:D). In a different greenhouse compartment with the same conditions, C. glomerata adult wasps were provided with first instar (L1) P. brassicae caterpillars and allowed to oviposit for 10-15 min. After oviposition, parasitised caterpillars were reared under the same conditions as unparasitised caterpillars. Parasitoid cocoons were collected a few days after they egressed from the caterpillar and placed in petridishes (90 mm) until emergence. Adult parasitoids were kept in a fine mesh cage and provided with honey and water. The hyperparasitoid B. galactopus was reared on P. brassicae caterpillars that had been parasitised by C. glomerata. To this end, to a Petri dish (90 mm) with seven parasitised L5 caterpillars, 15-20 B. galactopus adults were added. After egression of the parasitoid larvae from the caterpillar, the cocoons containing hyperparasitised C. glomerata were placed in a Petri dish (90 mm) in a climate chamber ($22 \pm 0.5^{\circ}$ C, 50-70% RH, and 16:8 h L:D) until adult B. galactopus emerged. Adult B. galactopus were kept in a rearing cage at room temperature, under natural light conditions (± 22°C, 35–45% RH) and were supplied with 5% honeywater.

Microbiome manipulation treatments

Previous research has shown that parasitism by *C. glomerata* alters the internal microbiome of *P. brassicae* caterpillars [41]. Therefore, half of the caterpillars studied were subjected to *C. glomerata* parasitism, while the other half remained unparasitised. When caterpillars had reached the fifth instar (L5), caterpillars were collected and subjected to different treatments. Each group of parasitised and unparasitised caterpillars was subjected to either (i) starvation (ST), (ii) starvation followed by a washing treatment to disrupt the external microbiome (ST+EMD) or (iii) were left untreated. Starvation was expected to affect the gut microbiome by exclusion of transient bacteria, but not the external microbiome. In addition, starvation was aimed at limiting the presence and effect of frass-related mVOCs. The starvation followed by external microbiome disruption was expected to affect the external microbiome in addition to the internal microbiome affected by the starvation treatment. Caterpillars undergoing starvation (ST and ST+EMD treatments) were starved for 24 hours by placing groups of ten caterpillars in a sterile plastic container with tissue paper (to absorb frass and moisture) during which they could empty their gut. Then those subjected to external microbiome disruption after starvation were washed three times with a 0.05% sodium hypochlorite solution followed by rinsing with

sterile water. We used a 0.05% sodium hypochlorite solution to disrupt the composition of the external microbiome of the caterpillars while preventing any mortality. This disruption reduces the absolute and relative abundance of most external microorganisms as a consequence of the different sensitivity to sodium hypochlorite and the differences in the adhesive capacity of bacteria [69,70]. Higher concentrations of sodium hypochlorite were proven to be more effective at removing bacteria but were too invasive for caterpillars. Only starved caterpillars were subjected to the external microbiome disruption treatment to limit excessive contaminations with bodily products (frass and regurgitate). Untreated caterpillars were also placed in groups of ten caterpillars in a sterile plastic container with tissue paper, but were supplied with fresh Brussels sprouts leaves and continued to feed, while other treatments were starved. Boxes with caterpillars had a pierced lid and were stored for 24 h at room temperature ($\pm 22^{\circ}$ C, 35–45% RH). Groups of eight caterpillars were subjected to headspace collection, followed by microbiome collection. In total every treatment was replicated ten times.

Caterpillar odour profile analysis

Headspace collection of caterpillar odours. To characterize the odour profiles of the starved, starved then external microbiome disrupted or untreated parasitised and unparasitised caterpillars, headspace samples were collected from ten cohorts of eight caterpillars per treatment, resulting in a total of 60 samples. However, one sample was (Sample 11, Pb-ST treatment) an extreme outlier based on the RDA analysis and was hence removed from the dataset. We followed the methodology described by [38] in which caterpillar headspace samples were collected in 500 mL glass jars sealed with a Viton-lined glass lid with an inlet and outlet for (clean) air. In our setup, each glass jar contained eight L5 P. brassicae caterpillars from the same treatment (Fig 1). A continuous flow of clean, synthetic air (Air Synthetic 4.0 Monitoring from Linde Gas, Schiedam, The Netherlands) was used as a carrier of volatiles at a flow rate of 100 mL min⁻¹. Volatiles emitted by the caterpillars were trapped by drawing air out of the glass jar at a suction rate of 100 mL min⁻¹ through a stainless-steel tube filled with 200 mg Tenax TA (20/35 mesh; CAMSCO, Houston, TX, USA) for 2 h. To prevent caterpillars from moving upward and blocking the air in- or outlets, each collection jar contained a restriction device made from a stainless-steel dome sieve (Cuisine elegance, $\emptyset = 8.5$ cm, mesh size = 12 mm) on a basis of aluminium foil (Fig 1). This device physically divided the setup into two to restrict caterpillar movement, but air could move freely through both parts. Directly after headspace collection, the Tenax TA cartridges were dry-purged under a stream of helium (50 ml min⁻¹) for 10 min at room temperature $(21 \pm 2^{\circ}C)$ to remove moisture before storage. In order to prevent any contribution from the collection set-up, the adsorbent material and the analytical system, we routinely trapped volatiles from empty jars containing a restriction device and included these as background samples.

Separation and detection of volatile compounds. The collected volatiles were thermally released from the Tenax TA adsorbent using an Ultra 50:50 thermal desorption unit (Markes, Llantrisant, Glamorgan, UK) at 250°C for 10 min under a helium flow of 20 mL min⁻¹. Simultaneously the volatiles were re-collected in a thermally cooled universal solvent trap: Unity (Markes) at 0°C. Once the desorption process was completed, volatile compounds were released from the cold trap by ballistic heating at 40°C s⁻¹ to 280°C. This was then kept for 10 min, while all the volatiles were transferred to a ZB-5MS analytical column (30 mL x 0.25 mm ID x 1 mm F.T.) with 10 m built-in guard column (Phenomenex, Torrance, CA, USA), placed inside the oven of a Thermo Trace GC Ultra (Thermo Fisher Scientific, Waltham, MA, USA), for further separation of the volatiles. The gas chromatograph oven temperature was initially

held at 40°C for 2 min and was immediately raised at 6°C min⁻¹ to a final temperature of 280°C, where it was kept for 4 min under a constant helium flow of 1 mL min⁻¹.

A Thermo Trace DSQ quadrupole mass spectrometer (Thermo Fisher Scientific) coupled to the gas chromatograph was operated in an electron impact ionisation (EI) mode at 70 eV in a full scan with a mass range of 35–400 amu at 4.70 scans s⁻¹. The MS transfer line and ion source were set at 275 and 250 °C, respectively. Automated baseline correction, peak selection (S/N > 3) and alignments of all extracted mass signals of the raw data were processed following an untargeted metabolomic workflow using MetAlign. This software uses the height of the peak after baseline correction to avoid problems with peak overlap, varying baselines, and peak tailing among others [71]. This produces detailed information on the relative abundance of mass signals representing the available metabolites. Next, the reconstruction of the extracted mass features into potential compounds was done using the MSClust software through data reduction by means of unsupervised clustering and extraction of putative metabolite mass spectra [72]. Tentative identification of volatile metabolites was based on comparison of the reconstructed mass spectra With those in the NIST 2008 and Wageningen Mass Spectral Database of Natural Products MS libraries, as well as experimentally obtained linear retention indices (LRI, Table 1).

Statistical analysis of volatile data. The volatile emission data as peak heights were imported to SIMCA-P 17 statistical software (Umetrics, Umea, Sweden), followed by log-transformation, mean-centering and unit-variance scaling before subjecting the data to multi-variate data analysis. Supervised orthogonal partial least squares-discriminant analysis (OPLS-DA) and unsupervised principal component analysis (PCA) statistical models were used as a tool to compare and correlate treatment groups. Whereas PCA was used when no significant model for OPLS-DA could be obtained. The results were visualised as score plots revealing the sample structure according to the model components, and loading plots showing the contribution of variables to the components as well as the relationship among the variables. R2 and Q2 metrics, which describe the explained variation within the data set and the predictability of the model, were calculated based on the averages of the sevenfold cross-validation. R2 and Q2 values range between 0 and 1, and the closer these metrics are to 1, the higher the variance explained by the model and the more reliable the predictive power of the model. Permutation and CV-ANOVA were used to validate models.

After identifying compounds with a variable importance in the projection (VIP) score of 1 or higher as potentially relevant, we subjected these to further analyses using a Kruskal-Wallis test followed by Dunn's test for multiple comparisons with a Bonferroni correction to determine significant differences between treatments. This was done using the dunn.test command in R-studio (dunn.test package) [73,74].

Caterpillar microbiome analysis

Microbiome collection and processing. The external and internal microbiome were immediately sampled after volatile collection, as described previously [41,64] (Fig 1). Briefly, to collect the external microbiome, caterpillars were put individually in a 2 mL microcentrifuge tube containing 1 mL of phosphate-buffered saline with 0.01% Tween80 (PBS-T), and vortexed for 20 seconds. The resulting solution then reflected the external microbiome of each individual. To remove any residual microorganisms, caterpillars were transferred into a new tube with 1 mL sodium hypochlorite (2.5%) and vortexed again for 20 seconds, followed by a final washing step in PBS-T. Each caterpillar was then dissected under sterile conditions to confirm presence or absence of *C. glomerata* larvae. Next, samples were pooled according to the same groups of eight caterpillars that were used for the volatile collection. To pool the

external microbiomes, the washing solution of the first caterpillar was centrifuged (13000 \times g, 10 min, room temperature), followed by removal of the supernatant. Subsequently, the washing solution of the second caterpillar of the same group was added to the same tube and centrifuged, after which the supernatant was removed again. This process was carried out until each of the eight caterpillars was included. The resulting pellet was resuspended in 500 µL nuclease free water and subsequently used for DNA extraction. For the internal samples, the dissected caterpillars were aseptically combined in a sterile 100 mL glass beaker and mixed under sterile conditions with an immersion homogeniser (RIVIERA&BAR, PPM530) at maximum speed for 20 s. The obtained mixture was then transferred into a 15 mL Falcon tube, vortexed and used for DNA extraction.

Microbial community characterisation. For each pooled sample, genomic DNA was extracted from 500 µL sample material using the PowerPro Soil Kit (Qiagen, Hilden, Germany) following the manufacturer's instructions, with one modification: in the second step of the protocol the use of a vortex adapter was replaced by two cycles of 30 s in the Bead Ruptor Elite at a speed of 5.5 m s⁻¹. To confirm the absence of contamination from the kit reagents', two negative controls were included by replacing the sample with 500 μ L sterile, DNA-free water. Next, for each sample the hypervariable V4 region of the bacterial 16S rRNA gene was amplified using Illumina-barcoded versions of primers 515F and 806R [75], designed according to a dual index sequencing strategy [76] (Table C in S1 Supporting Information). Additionally, four negative controls in which template DNA was replaced by DNA-free water were included, as well as a DNA mock community sample, composed of a number of bacterial species that likely occur in or on insects [41] (Table D in S1 Supporting Information). PCR amplification, library preparation, sequencing and bioinformatics analysis were performed as described previously [41]. Sequences were classified into zero-radius operational taxonomic units (zOTUs, also known as amplicon sequence variants (ASVs) [77,78]), enabling to resolve sequence differences by as little as a single base pair. Further, to remove potential contaminants, the data set was decontaminated in R (v4.0.2) using microDecon (v1.2.0) [79] based on zOTU prevalence in the investigated samples versus the mean of the four PCR controls [74,80]. Additionally, in accordance with the results obtained for the mock community, zOTUs occurring below a 0.3% relative abundance threshold per sample were discarded from further analysis. Finally, the number of sequences was rarefied to 1,000 sequences per sample, representing the least number of high-quality sequences obtained per sample. The taxonomic origin of each zOTU was determined with the SINTAX algorithm as implemented in USEARCH based on the SILVA Living Tree Project v1.23 (Table E in S1 Supporting Information). The identity of the most important zOTUs was further verified with a BLAST search in GenBank against type materials. When no significant similarity was found with type materials, the BLAST search was performed against entire GenBank. The negative DNA extraction samples showed no visible band after the amplification, therefore no contaminants were present in the DNA extractions. Analysis of the mock community demonstrated that only the expected taxa were found, indicating that the experimental conditions were met to achieve robust data. The sequences obtained in this study were deposited in the Sequence Read Archive (SRA) at NCBI under Bioproject PRJNA878850.

Statistics on microbiome data. For each sample, a rarefaction curve was generated using the Phyloseq package in R showing the number of observed zOTUs as a function of the number of sequences [74,81]. For all samples, rarefaction curves approached saturation, indicating that our sequencing depth was sufficient to cover the microbial diversity (S7 Fig). Phyloseq package in R was used as well to calculate observed zOTU richness and Shannon diversity. A three-way analysis of variance (ANOVA) was used to assess whether health status (parasitised or unparasitised), origin (external or internal microbiome) and treatment affected zOTU

richness and Shannon diversity. All two-way interactions and three-way interactions were included in the model as well. A further analysis of variance was conducted on external and internal samples separately. The bacterial community composition was visualised by non-metric multidimensional scaling (NMDS) using the Bray-Curtis coefficient as distance measure in the R software package Vegan [82]. The coefficient was based on the Hellinger transformed relative abundance data of the observed bacteria in each sample. To test whether bacterial communities differed between health status, origin and treatments, the "adonis" function in the software package Vegan [82] was used to perform a permutational analysis of variance (PERMANOVA) [83]. All factors and their interactions were included as fixed factors in the analysis. Significance was tested using 1,000 permutations.

Forward selection RDA to link microbiome with volatile data

To investigate whether the volatile composition of parasitised and unparasitised caterpillars was significantly related to variation in microbiome, redundancy analyses using the rda function in Vegan [80] were used. In these analyses, log transformed peak heights were used as dependent variables, whereas zOTU relative abundance data were used as independent data. A forward selection procedure using the ordistep function in Vegan was applied to retain those zOTUs that had a significant impact on volatile composition. Significance of the final model was tested using the anova.cca function in Vegan [80] using 1000 permutations. Finally, triplots were constructed, which allow to visualise the relationship between samples, volatiles and the selected zOTUs. Data were analysed separately for the external and internal microbiome.

Behavioural assays with hyperparasitoids

Hyperparasitoid preference for caterpillar odours. The response of mated *B. galactopus* females to the odours of differently treated unparasitised or parasitised *P. brassicae* was tested in a Y-tube olfactometer. A similar setup was successfully used to study the preference of the parasitoid wasp *Nasonia vitripennis* (Hymenoptera: Pteromalidae) to odours of conspecifics [84]. A glass Y-tube with an 8.5 cm stem, two 5 cm arms and a diameter of 0.9 cm (Fig 8A) was tilted upward on a white wooden board with an angle of 40° towards a single light source (TL-D 58 W, Phillips, the Netherlands) (Fig 8B). A fine gauze mesh was placed over each end of the Y-tube to create a physical barrier, then a 5 mL transparent pipette tip containing the test caterpillars was mounted on each end of the Y-tube. To test *B. galactopus*' preference for *C. glomerata* odours, two fifth instar *P. brassicae* caterpillars were placed in a pipette tip (Fig 8A). For the preference of *H. ebeninus* odours this were four third instar *P. brassicae* caterpillars because *H. ebeninus* restricts the growth of its host [85]. A charcoal-filtered airflow of 500 mL min⁻¹ was split and led through both arms, starting at the narrow end of the pipette tip and was sealed with Teflon tape (Fig 8A).

Prior to each set of choices, caterpillar odours were allowed to diffuse for 15 min from the pipette tips containing caterpillars into the arms of the Y-tube olfactometer. If present, frass droppings were removed prior to testing. Tests started when one *B. galactopus* female was released at the base of the Y-tube from a glass vial. A choice was noted when the hyperparasitoid passed a drawn line at 1 cm from the end of either arm for 15 seconds. If 10 min had passed without the hyperparasitoid making a choice, this was recorded as a no-choice and excluded from statistical analysis. Each female was used only once and a total of 10 females per day was used per tested combination. After every 5 tested females the Y-tube was turned by 180 degrees to compensate for any unforeseen asymmetry bias. We subjected unparasitised and parasitised *P. brassicae* to the same treatments as mentioned before. Untreated caterpillars were omitted from choice-assays to prevent frass production from contaminating the setup





https://doi.org/10.1371/journal.ppat.1011262.g008

and overruling the effect of caterpillar body odours. The preference of *B. galactopus* for caterpillar odours was analysed using two-tailed binomial tests with the binom.test command in R-studio [74].

Hyperparasitoid preference for caterpillar frass odours

We used a two-chamber olfactometer setup [38] to assess the attractiveness of frass odours from (parasitised) caterpillars to the true hyperparasitoid *B. galactopus*. Three treatments were included, an empty chamber (E), a chamber containing a sample of 10 pellets of frass produced by an unparasitised L5 *P. brassicae* caterpillar (Pb) or by an L5 *P. brassicae* caterpillar parasitised by *C. glomerata* (Cg). For each treatment combination, 71 replicates in total were included. In order to maintain the frass equally fresh, a filter paper moisturised with sterile Milli-Q water was placed at the bottom of all chambers, including empty ones. Five mated female *B. galactopus* wasps were released inside the experimental arena containing the two chambers. After exactly one hour, the number of wasps present in each chamber was assessed for each combination. When no wasps were found inside any of the two chambers at the end

of the bioassay, the cohort was considered unresponsive and was excluded from the analysis. Bioassays of the three treatment combinations were always performed during the same time, in order to account for variation caused by hour and day. The bioassays were conducted between 10.00 and 16.00h. After each bioassay, the chambers and parts of the set up in contact with frass were wiped with 70% ethanol, dried with tissue paper and the whole set up was flushed with high-pressure clean air. A cohort of 5 wasps was considered as a replicate. Thus for each of the combinations we ended up with two distributions of scores from 0 to 5. We compared the two non-parametric distributions and their median with Mann-Whitney U tests separately for each treatment combination in R-studio using the wilcox.test command (unpaired) [74].

Hyperparasitoid behavioural responses to caterpillars injected with *Wolbachia*

Caterpillars were injected according to the method described in [86] with modifications. Wolbachia was extracted from maturing C. glomerata larvae instead of adult parasitoids. Because Wolbachia in adult parasitoids co-occurs with abundant polydnaviruses in the calyx fluid of the parasitoid ovaries, extraction of Wolbachia from the calyx would lead to contamination with polydnavirus that is known to affect the caterpillar's phenotype [87]. Previous research has shown that the internal microbiome of C. glomerata larvae is dominated by Wolbachia, up to 100% relative abundance [41] and do not contain polydnavirus particles. After anaesthesia, L5 P. brassicae caterpillars parasitised by C. glomerata were aseptically dissected and the parasitoid larvae were collected and surface-sterilized as described in [41]. Parasitoid larvae were crushed in a centrifuge tube with a small pestle in 20 µL sterile PBS and subsequently homogenized in 800 µL PBS with two glass beads (2 mm diameter) using a TissueLyser II (Qiagen, Hilden, Germany; 3 min at 30 Hz). Following a 10-min centrifugation at 2500 g, the supernatant was collected and filtered through a Puradisc 13 glass microfiber syringe filter (pore size: 2.7 µm; Whatman, Amersham, UK). The filtered solution was then centrifuged at 18500 g for 5 min to pellet the bacteria. The cell pellet was re-suspended in 10 µL sterile PBS and used for the injections. Parasitoid larvae were processed in groups of 20 individuals to obtain Wolbachia densities similar to those found in 10 ovary pairs of C. glomerata. The concentration of *Wolbachia* per ovary pair was determined at 2×10^2 16S rRNA gene copies per µL DNA by qPCR; for primers and PCR conditions, see [88]. Caterpillars were anaesthetized with CO₂ and cell suspensions were injected into P. brassicae caterpillars (right behind the head) using a Femtojet microinjector (Eppendorf, Hamburg, Germany). Injections were performed with aliquots of 0.1 μ L, assuming that 1/10 of calvx fluid is injected with the eggs during a parasitism bout [86].

The following treatments were included: (1) unparasitised caterpillars injected twice with PBS (0.1 μ L) (Unparasitised-PBS; negative control); (2) parasitised caterpillars injected twice with PBS (Parasitised-PBS; positive control); (3) unparasitised caterpillars injected first with *Wolbachia* and then with PBS (Wolb-PBS); and (4) unparasitised caterpillars injected twice with *Wolbachia* (Wolb-Wolb). The first injection was performed on first instar caterpillars, while the second injection was performed the day before performing the behavioural assay with caterpillars at the 5th instar. Caterpillars were maintained in separate cages per treatment under the previous mentioned conditions until they were used in the behavioural assays.

Behavioural assays were performed when caterpillars had reached the fifth instar as described in [38]. One caterpillar and one mated naïve female of *B. galactopus* were placed in a clean 9 cm diameter Petri dish (1.9 cm height), after which the hyperparasitoid's behaviour was monitored. Over one hour, we recorded the time it took for *B. galactopus* to make first

contact and mount the caterpillar. Next, in the hour following mounting, the total time the hyperparasitoid spent on top of the caterpillar was recorded. If the hyperparasitoid did not mount on the caterpillar within the period of one hour, it was considered as a non-responder and removed from the analysis. Furthermore, to ensure that the positive control only consisted of parasitised caterpillars, parasitised individuals were dissected in order to confirm parasitism, and caterpillars without parasitoid larvae were discarded from the analysis. In total, the experiment was performed with 39 caterpillars for the unparasitised-PBS group, 36 for Parasitised-PBS, 50 for Wolb-PBS, and 46 for Wolb-Wolb. To investigate whether the time used to make contact and total mounting differed between treatment, a non-parametric Analysis of Variance was used. Subsequently, post-hoc comparisons using Dunn's test were performed to see which treatment combinations differed significantly from each other. P-values were adjusted for multiple comparisons using the Hochberg method.

Supporting information

S1 Fig. Overview of caterpillar-associated bacterial communities and volatiles of starved then external microbiome disrupted unparasitised- or *C. glomerata* parasitised caterpillars (*Pieris brassicae*). a) NMDS ordination plots based on Bray–Curtis distances of Hellinger-transformed relative abundance data of the external (dermal) bacterial communities. b) NMDS ordination for the internal bacterial communites. c) OPLS-DA plot for the volatile blends of different groups of caterpillars. The Hotelling's T2 ellipse confines the confidence region (95%) of the score plot. d) The loading plot defining the contribution of each of the volatile compound to the separation of treatment groups. Volatiles compounds closer to a treatment in the plot are more correlated. For compound identity see <u>Table 1</u>. Abbreviations used: Cg-ST+EMD = starved then external microbiome disrupted unparasitised caterpillars. (TIF)

S2 Fig. Overview of caterpillar-associated bacterial communities and volatiles of starved unparasitised- or *C. glomerata* parasitised caterpillars (*Pieris brassicae*). a) NMDS ordination plots based on Bray–Curtis distances of Hellinger-transformed relative abundance data of the external (dermal) bacterial communities. b) NMDS ordination for the internal bacterial communities. c) OPLS-DA plot for the volatile blends of different groups of caterpillars. The Hotelling's T2 ellipse confines the confidence region (95%) of the score plot. d) The loading plot defining the contribution of each of the volatile compound to the orientation of sample groups in the score plot. Volatiles compounds closer to a treatment in the plot are more correlated. For compound identity see Table 1. Abbreviations used: Cg-ST = starved parasitised caterpillar. (TIF)

S3 Fig. Overview of caterpillar-associated bacterial communities and volatiles of starved then external microbiome disrupted or only starved *C. glomerata* parasitised caterpillars (*Pieris brassicae*). a) NMDS ordination plots based on Bray–Curtis distances of Hellinger-transformed relative abundance data of the external (dermal) bacterial communities. b) NMDS ordination for the internal bacterial communites. c) PCA plot for the volatile blends of different groups of caterpillars. The Hotelling's T2 ellipse confines the confidence region (95%) of the score plot. d) The loading plot defining the contribution of each of the volatile compound to the orientation of sample groups in the score plot. For compound identity see Table 1. Abbreviations used: Cg-ST = starved parasitised caterpillar; Cg-ST+EMD = starved

then external microbiome disrupted parasitised caterpillar. (TIF)

S4 Fig. Overview of caterpillar-associated bacterial communities and volatiles of starved then external microbiome disrupted or starved unparasitised caterpillars (*Pieris brassicae***).** a) NMDS ordination plots based on Bray–Curtis distances of Hellinger-transformed relative abundance data of the external (dermal) bacterial communities (stress = 0.132). b) NMDS ordination for the internal bacterial communites (stress = 0.115). c) PCA plot for the volatile blends of different groups of caterpillars. The Hotelling's T2 ellipse confines the confidence region (95%) of the score plot. d) The loading plot defining the contribution of each of the volatile compound to the first two principal components. For compound identity see <u>Table 1</u>. Abbreviations used: Pb-ST = starved unparasitised caterpillar; Pb-ST+EMD = starved then external microbiome disrupted unparasitised caterpillar. (TIF)

S5 Fig. *B. galactopus* **preference for frass odours.** The number of *B. galactopus* found inside a chamber of the two-chamber olfactometer after one hour (out of 5 individuals). Chambers were either empty or contained fresh frass from L5 unparasitised or parasitised caterpillars. *N* = the number of replicates, for each replicate 5 *B. galactopus* were released in the olfactometer. *P* was calculated using a Mann-Whitney U-test. (TIF)

S6 Fig. Preference of hyperparasitoids for caterpillar body odours to healthy and parasitised caterpillars in the L3 stage. For this experiment caterpillars were parasitised by the parasitoid *Hyposoter ebeninus*. Preference of the hyperparasitoid *Baryscapus galactopus* for caterpillar body odours was tested with Y-tube olfactometer tests. Tested combinations were selected according to the hypothesis that hyperparasitoids can use caterpillar body odours, which are (at least partially) determined by the external microbiome. Numbers between brackets indicate the number of wasps that made a choice within 10 min from the start of the experiment versus the total number of wasps tested. *** P < 0.001; two-sided binomial test. Abbreviations used: AIR = clean air; HE = Untreated parasitised caterpillar; HE-ST = starved parasitised caterpillar; HE-ST+EMD = starved then external microbiome disrupted parasitised caterpillar. Pb = Untreated unparasitised caterpillar; Pb-ST = starved unparasitised caterpillar; Pb-ST+EMD = starved and external microbiome disrupted unparasitised caterpillar; (TIF)

S7 Fig. Overview of the diversity of caterpillar-associated bacterial communities of untreated, starved or starved then external microbiome disrupted *C. glomerata* parasitised or unparasitised caterpillars (*Pieris brassicae*). a) The shannon diversity of the external microbiome b) The Shannon diversity of the internal microbiome. (TIF)

S8 Fig. Rarefaction curves for the different samples studied, based on the bacterial V4 dataset. Rarefaction curves approached saturation, indicating that our sequencing depth was sufficient to cover the microbial diversity. (TIF)

S1 Supporting Information. Table A: VIP scores for the volatile compounds in pairwise comparisons of untreated, starved then external microbiome disrupted or starved unparasitised- or *C. glomerata* parasitised caterpillars (*Pieris brassicae*). Bold face VIP scores are higher than 1 and indicate the most influential VOC for separation of the mentioned treatments. Compounds and pairwise comparisons in green are correlated with the presence/

absence of frass. Compounds and pairwise comparisons in yellow are correlated with parasitism status. Table B: Full overview of caterpillar-associated bacterial communities of untreated, starved then external microbiome disrupted or starved *C. glomerata* parasitised or unparasitised caterpillars (*P. brassicae*). Table C: Primer design and sample-specific barcodes. Table D: Composition of mock community. Table E: Identification of bacterial zero radius operational taxonomic units (ZOTUs) according to the Silva v1.23 database and distribution over the investigated samples. (XLSX)

Acknowledgments

We thank Pieter Rouweler, Arnout Berendsen, the late Frans van Aggelen, and André Gidding for rearing *P. brassicae* and *C. glomerata*. Further, we are grateful to Bram Kamps and Maximilien Cuny for rearing *H. ebeninus*, Janneke Bloem and Filippo Guerra for assistance in the lab.

Author Contributions

Conceptualization: Mitchel E. Bourne, Gabriele Gloder, Marijn Slingerland, Bart Lievens, Hans Jacquemyn, Marcel Dicke, Erik H. Poelman.

- Data curation: Mitchel E. Bourne, Gabriele Gloder, Sam Crauwels, Bart Lievens.
- Formal analysis: Mitchel E. Bourne, Gabriele Gloder, Berhane T. Weldegergis, Andrea Ceribelli, Sam Crauwels, Hans Jacquemyn.
- Funding acquisition: Bart Lievens, Hans Jacquemyn, Marcel Dicke.
- Investigation: Mitchel E. Bourne, Gabriele Gloder, Berhane T. Weldegergis, Marijn Slingerland, Andrea Ceribelli.
- Methodology: Mitchel E. Bourne, Gabriele Gloder, Berhane T. Weldegergis, Marijn Slingerland, Andrea Ceribelli.
- Project administration: Bart Lievens, Marcel Dicke.

Resources: Marcel Dicke.

Software: Sam Crauwels.

Supervision: Bart Lievens, Hans Jacquemyn, Marcel Dicke, Erik H. Poelman.

Validation: Mitchel E. Bourne, Berhane T. Weldegergis, Marcel Dicke, Erik H. Poelman.

- Visualization: Mitchel E. Bourne, Gabriele Gloder, Berhane T. Weldegergis, Andrea Ceribelli, Hans Jacquemyn.
- Writing original draft: Mitchel E. Bourne, Gabriele Gloder, Erik H. Poelman.
- Writing review & editing: Mitchel E. Bourne, Gabriele Gloder, Berhane T. Weldegergis, Marijn Slingerland, Andrea Ceribelli, Sam Crauwels, Bart Lievens, Hans Jacquemyn, Marcel Dicke, Erik H. Poelman.

References

- Zilber-Rosenberg I, Rosenberg E. Role of microorganisms in the evolution of animals and plants: The hologenome theory of evolution. FEMS Microbiol Rev. 2008; 32: 723–735. https://doi.org/10.1111/j. 1574-6976.2008.00123.x PMID: 18549407
- Rosenberg E, Zilber-Rosenberg I. The hologenome concept of evolution after 10 years. Microbiome. 2018; 6: 1–14. https://doi.org/10.1186/s40168-018-0457-9 PMID: 29695294

- Margulis L. Symbiogenesis and symbionticism. In: Margulis L, Fester R, editors. Symbiosis as a source of evolutionary innovation: speciation and morphogenesis. Cambridge, MA: MIT Press; 1991. pp. 1– 14.
- 4. Mindell DP. Phylogenetic consequences of symbioses: Eukarya and Eubacteria are not monophyletic taxa. Biosystems. 1992; 27: 53–62. https://doi.org/10.1016/0303-2647(92)90046-2 PMID: 1391691
- Bordenstein SR, Theis KR. Host biology in light of the microbiome: ten principles of holobionts and hologenomes. PLoS Biol. 2015; 13: e1002226. <u>https://doi.org/10.1371/journal.pbio.1002226</u> PMID: 26284777
- 6. Dawkins R. The extended phenotype. Oxford, UK: Oxford Univ. Press; 1982.
- Engel P, Moran NA. The gut microbiota of insects—diversity in structure and function. FEMS Microbiol Rev. 2013; 37: 699–735. https://doi.org/10.1111/1574-6976.12025 PMID: 23692388
- Frago E, Zytynska SE, Fatouros NE. Microbial symbionts of herbivorous species across the insect tree. Advances in Insect Physiology. Elsevier Ltd.; 2020. pp. 111–159. https://doi.org/10.1016/bs.aiip.2020. 04.002
- Weisskopf L, Schulz S, Garbeva P. Microbial volatile organic compounds in intra-kingdom and interkingdom interactions. Nat Rev Microbiol. 2021; 19: 391–404. https://doi.org/10.1038/s41579-020-00508-1 PMID: 33526910
- Archie EA, Theis KR. Animal behaviour meets microbial ecology. Anim Behav. 2011; 82: 425–436. https://doi.org/10.1016/j.anbehav.2011.05.029
- Rering CC, Beck JJ, Hall GW, McCartney MM, Vannette RL. Nectar-inhabiting microorganisms influence nectar volatile composition and attractiveness to a generalist pollinator. New Phytol. 2018; 220: 750–759. https://doi.org/10.1111/nph.14809 PMID: 28960308
- Sharon G, Segal D, Ringo JM, Hefetz A, Zilber-Rosenberg I, Rosenberg E. Commensal bacteria play a role in mating preference of *Drosophila melanogaster*. Proc Natl Acad Sci U S A. 2010; 107: 20051– 20056. https://doi.org/10.1073/pnas.1009906107 PMID: 21041648
- Verhulst NO, Qiu YT, Beijleveld H, Maliepaard C, Knights D, Schulz S, et al. Composition of human skin microbiota affects attractiveness to malaria mosquitoes. PLoS One. 2011; 6: e28991. https://doi.org/10. 1371/journal.pone.0028991 PMID: 22216154
- Goelen T, Sobhy IS, Vanderaa C, de Boer JG, Delvigne F, Francis F, et al. Volatiles of bacteria associated with parasitoid habitats elicit distinct olfactory responses in an aphid parasitoid and its hyperparasitoid. Funct Ecol. 2020; 34: 507–520. https://doi.org/10.1111/1365-2435.13503
- Mazorra-Alonso M, Tomás G, Soler JJ. Microbially mediated chemical ecology of animals: A review of its role in conspecific communication, parasitism and predation. Biology. 2021; 10: 274. <u>https://doi.org/ 10.3390/biology10040274</u> PMID: 33801728
- Busula AO, Takken W, de Boer JG, Mukabana WR, Verhulst NO. Variation in host preferences of malaria mosquitoes is mediated by skin bacterial volatiles. Med Vet Entomol. 2017; 31: 320–326. https://doi.org/10.1111/mve.12242 PMID: 28639690
- Verhulst NO, Umanets A, Weldegergis BT, Maas JPA, Visser TM, Dicke M, et al. Do apes smell like humans? The role of skin bacteria and volatiles of primates in mosquito host selection. J Exp Biol. 2018; 221: 1–11. https://doi.org/10.1242/jeb.185959 PMID: 30297513
- Calcagnile M, Tredici SM, Talà A, Alifano P. Bacterial semiochemicals and transkingdom interactions with insects and plants. Insects. 2019; 10: 1–30. <u>https://doi.org/10.3390/insects10120441</u> PMID: 31817999
- Showering A, Martinez J, Benavente ED, Gezan SA, Jones RT, Oke C, et al. Skin microbiome alters attractiveness to *Anopheles* mosquitoes. BMC Microbiol. 2022; 22: 1–14. <u>https://doi.org/10.1186/s12866-022-02502-4</u> PMID: 35410125
- van Neerbos FAC, Dewitte P, Wäckers F, Wenseleers T, Jacquemyn H, Lievens B. Bacterial volatiles elicit differential olfactory responses in insect species from the same and different trophic levels. Insect Sci. 2023; 1–23. https://doi.org/10.1111/1744-7917.13176 PMID: 36644938
- Verhulst NO, Mukabana WR, Takken W, Smallegange RC. Human skin microbiota and their volatiles as odour baits for the malaria mosquito *Anopheles gambiae* s.s. Entomol Exp Appl. 2011; 139: 170– 179. https://doi.org/10.1111/j.1570-7458.2011.01119.x
- Revadi S V., Giannuzzi VA, Vetukuri RR, Walker WB, Becher PG. Larval response to frass and guaiacol: detection of an attractant produced by bacteria from *Spodoptera littoralis* frass. J Pest Sci. 2021; 94: 1105–1118. https://doi.org/10.1007/s10340-021-01352-9
- Agelopoulos NG, Dicke M, Posthumus MA. Role of volatile inforchemicals emitted by feces of larvae in host-searching behavior of parasitoid *Cotesia rubecula* (Hymenoptera: Braconidae): A behavioral and chemical study. J Chem Ecol. 1995; 21: 1789–1811. https://doi.org/10.1007/BF02033677 PMID: 24233830

- Chuche J, Xuéreb A, Thiéry D. Attraction of *Dibrachys cavus* (Hymenoptera: Pteromalidae) to its host frass volatiles. J Chem Ecol. 2006; 32: 2721–2731. https://doi.org/10.1007/s10886-006-9195-8 PMID: 17091401
- Desneux N, Ramírez-Romero R, Bokonon-Ganta AH, Bernal JS. Attraction of the parasitoid *Cotesia* marginiventris to host (*Spodoptera frugiperda*) frass is affected by transgenic maize. Ecotoxicology. 2010; 19: 1183–1192. https://doi.org/10.1007/s10646-010-0502-3 PMID: 20480228
- Gould AL, Zhang V, Lamberti L, Jones EW, Obadia B, Korasidis N, et al. Microbiome interactions shape host fitness. Proc Natl Acad Sci. 2018; 115: E11951–E11960. <u>https://doi.org/10.1073/pnas.</u> 1809349115 PMID: 30510004
- Negroni MA, Segers FHID, Vogelweith F, Foitzik S. Immune challenge reduces gut microbial diversity and triggers fertility-dependent gene expression changes in a social insect. BMC Genomics. 2020; 21: 1–16. https://doi.org/10.1186/s12864-020-07191-9 PMID: 33225893
- Fredensborg BL, Fossdal í Kálvalíð I, Johannesen TB, Stensvold CR, Nielsen H V., Kapel CMO. Parasites modulate the gut-microbiome in insects: A proof-of-concept study. PLoS One. 2020; 15: e0227561. https://doi.org/10.1371/journal.pone.0227561 PMID: 31935259
- 29. Dicke M, Cusumano A, Poelman EH. Microbial symbionts of parasitoids. Annu Rev Entomol. 2020; 65: 171–190. https://doi.org/10.1146/annurev-ento-011019-024939 PMID: 31589823
- 30. Godfray HCJ. Parasitoids: behavioral and evolutionary ecology. Princeton, NJ: Princeton Press; 1994.
- Cuny MAC, Poelman EH. Evolution of koinobiont parasitoid host regulation and consequences for indirect plant defence. Evol Ecol. 2022; 36: 299–319. https://doi.org/10.1007/s10682-022-10180-x PMID: 35663232
- Tan C-W, Peiffer M, Hoover K, Rosa C, Acevedo FE, Felton GW. Symbiotic polydnavirus of a parasite manipulates caterpillar and plant immunity. Proc Natl Acad Sci. 2018; 115: 5199–5204. <u>https://doi.org/ 10.1073/pnas.1717934115</u> PMID: 29712862
- Wang Y, Wu X, Wang Z, Chen T, Zhou S, Chen J, et al. Symbiotic bracovirus of a parasite manipulates host lipid metabolism via tachykinin signaling. PLoS Pathog. 2021; 17: e1009365. <u>https://doi.org/10. 1371/journal.ppat.1009365</u> PMID: 33647060
- Cavichiolli de Oliveira N, Cônsoli FL. Beyond host regulation: Changes in gut microbiome of permissive and non-permissive hosts following parasitization by the wasp *Cotesia flavipes*. FEMS Microbiol Ecol. 2020; 96: 1–17. https://doi.org/10.1093/femsec/fiz206 PMID: 31860060
- Wang J, Mason CJ, Ju X, Xue R, Tong L, Peiffer M, et al. Parasitoid causes cascading effects on plantinduced defenses mediated through the gut bacteria of host caterpillars. Front Microbiol. 2021; 12: 1– 11. https://doi.org/10.3389/fmicb.2021.708990 PMID: 34552570
- Hammer TJ, Janzen DH, Hallwachs W, Jaffe SP, Fierer N. Caterpillars lack a resident gut microbiome. Proc Natl Acad Sci. 2017; 114: 9641–9646. https://doi.org/10.1073/pnas.1707186114 PMID: 28830993
- Hannula SE, Zhu F, Heinen R, Bezemer TM. Foliar-feeding insects acquire microbiomes from the soil rather than the host plant. Nat Commun. 2019; 10: 1–9. <u>https://doi.org/10.1038/s41467-019-09284-w</u> PMID: 30890706
- Zhu F, Weldegergis BT, Lhie B, Harvey JA, Dicke M, Poelman EH. Body odors of parasitized caterpillars give away the presence of parasitoid larvae to their primary hyperparasitoid enemies. J Chem Ecol. 2014; 40: 986–995. https://doi.org/10.1007/s10886-014-0500-7 PMID: 25236382
- Harvey JA, Gols R, Vet LEM, Kruidhof HM. Development of a hyperparasitoid wasp in different stages of its primary parasitoid and secondary herbivore hosts. J Insect Physiol. 2012; 58: 1463–1468. <u>https:// doi.org/10.1016/i.jinsphys.2012.08.013 PMID: 22939901</u>
- 40. Poelman EH, Cusumano A, de Boer JG. The ecology of hyperparasitoids. Annu Rev Entomol. 2022; 67: 143–161. https://doi.org/10.1146/annurev-ento-060921-072718 PMID: 34606363
- Gloder G, Bourne ME, Verreth C, Wilberts L, Bossaert S, Crauwels S, et al. Parasitism by endoparasitoid wasps alters the internal but not the external microbiome in host caterpillars. Anim Microbiome. 2021; 3: 1–15. https://doi.org/10.1186/s42523-021-00135-y PMID: 34654483
- Silva DB, Weldegergis BT, Van Loon JJA, Bueno VHP. Qualitative and quantitative differences in herbivore-induced plant volatile blends from tomato plants infested by either *Tuta absoluta* or *Bemisia tabaci*. J Chem Ecol. 2017; 43: 53–65. https://doi.org/10.1007/s10886-016-0807-7 PMID: 28050733
- Schettino M, Grasso DA, Weldegergis BT, Castracani C, Mori A, Dicke M, et al. Response of a predatory ant to volatiles emitted by aphid- and caterpillar-infested cucumber and potato plants. J Chem Ecol. 2017; 43: 1007–1022. https://doi.org/10.1007/s10886-017-0887-z PMID: 28951999
- 44. Manrique G, Vitta ACR, Ferreira RA, Zani CL, Unelius CR, Lazzari CR, et al. Chemical communication in chagas disease vectors. Source, identity, and potential function of volatiles released by the metasternal and Brindley's glands of *Triatoma infestans* adults. J Chem Ecol. 2006; 32: 2035–2052. https://doi. org/10.1007/s10886-006-9127-7 PMID: 16902820

- 45. Witte V, Disney RHL, Weissflog A, Maschwitz U. Studies in European ant-decapitating flies (Diptera: Phoridae): Ant alarm pheromone as host finding cue in *Pseudacteon brevicauda*, a parasite of *Myrmica rubra* (Formicidae: Myrmicinae). J Nat Hist. 2010; 44: 905–912. <u>https://doi.org/10.1080/</u>00222930903437366
- 46. Sun Q, Haynes KF, Zhou X. Dynamic changes in death cues modulate risks and rewards of corpse management in a social insect. Funct Ecol. 2017; 31: 697–706. <u>https://doi.org/10.1111/1365-2435.12754</u>
- Viklund L, Bång J, Schroeder M, Hedenström E. Identification of male produced compounds in the bark beetle *Polygraphus subopacus* and establishment of (Z)-2-(3,3-dimethylcyclohexylidene)-ethanol as an aggregation pheromone component. Chemoecology. 2021; 31: 367–376. <u>https://doi.org/10.1007/ s00049-021-00358-0</u>
- Xie J, Liu T, Yi C, Liu X, Tang H, Sun Y, et al. Antenna-biased odorant receptor HvarOR25 in *Hippoda-mia variegata* tuned to allelochemicals from hosts and habitat involved in perceiving preys. J Agric Food Chem. 2022; 70: 1090–1100. https://doi.org/10.1021/acs.jafc.1c05593 PMID: 35072468
- 49. Zhu F, Broekgaarden C, Weldegergis BT, Harvey JA, Vosman B, Dicke M, et al. Parasitism overrides herbivore identity allowing hyperparasitoids to locate their parasitoid host using herbivore-induced plant volatiles. Mol Ecol. 2015; 24: 2886–2899. https://doi.org/10.1111/mec.13164 PMID: 25789566
- Lemfack MC, Gohlke B-O, Toguem SMT, Preissner S, Piechulla B, Preissner R. mVOC 2.0: a database of microbial volatiles. Nucleic Acids Res. 2018; 46: D1261–D1265. https://doi.org/10.1093/nar/gkx1016 PMID: 29106611
- Schulz S, Dickschat JS. Bacterial volatiles: the smell of small organisms. Nat Prod Rep. 2007; 24: 814– 842. https://doi.org/10.1039/b507392h PMID: 17653361
- Moisan K, Cordovez V, van de Zande EM, Raaijmakers JM, Dicke M, Lucas-Barbosa D. Volatiles of pathogenic and non-pathogenic soil-borne fungi affect plant development and resistance to insects. Oecologia. 2019; 190: 589–604. https://doi.org/10.1007/s00442-019-04433-w PMID: 31201518
- Paniagua Voirol LR, Weinhold A, Johnston PR, Fatouros NE, Hilker M. Legacy of a butterfly's parental microbiome in offspring performance. Appl Environ Microbiol. 2020; 86: 1–16. <u>https://doi.org/10.1128/</u> AEM.00596-20 PMID: 32276976
- Phalnikar K, Kunte K, Agashe D. Dietary and developmental shifts in butterfly-associated bacterial communities. R Soc Open Sci. 2018; 5: 1–18. https://doi.org/10.1098/rsos.171559 PMID: 29892359
- Luo Y, Zhou M, Zhao Q, Wang F, Gao J, Sheng H, et al. Complete genome sequence of *Sphingomonas* sp. Cra20, a drought resistant and plant growth promoting rhizobacteria. Genomics. 2020; 112: 3648– 3657. https://doi.org/10.1016/j.ygeno.2020.04.013 PMID: 32334112
- 56. Van Nouhuys S, Kohonen M, Duplouy A. Wolbachia increases the susceptibility of a parasitoid wasp to hyperparasitism. J Exp Biol. 2016; 219: 2984–2990. <u>https://doi.org/10.1242/jeb.140699</u> PMID: 27707863
- Furihata S, Hirata M, Matsumoto H, Hayakawa Y. Bacteria endosymbiont, *Wolbachia*, promotes parasitism of parasitoid wasp Asobara japonica. PLoS One. 2015; 10: 1–16. <u>https://doi.org/10.1371/journal.pone.0140914</u> PMID: 26492411
- Bi J, Wang Y. The effect of the endosymbiont *Wolbachia* on the behavior of insect hosts. Insect Sci. 2020; 27: 846–858. https://doi.org/10.1111/1744-7917.12731 PMID: 31631529
- 59. Richard F-J. Symbiotic bacteria influence the odor and mating preference of their hosts. Front Ecol Evol. 2017; 5: 1–9. https://doi.org/10.3389/fevo.2017.00143
- O'Neill SL M. Pettigrew M, Sinkins SP, Braig HR, Andreadis TG, Tesh RB. In vitro cultivation of Wolbachia pipientis in an Aedes albopictus cell line. Insect Mol Biol. 1997; 6: 33–39. https://doi.org/10.1046/j. 1365-2583.1997.00157.x PMID: 9013253
- Lutz HL, Gilbert JA, Dick CW. Associations between Afrotropical bats, eukaryotic parasites, and microbial symbionts. Mol Ecol. 2022; 31: 1939–1950. https://doi.org/10.1111/mec.16044 PMID: 34181795
- Grasswitz TR. Contact kairomones mediating the foraging behavior of the aphid hyperparasitoid Alloxysta victrix (Westwood) (Hymenoptera: Charipidae). J Insect Behav. 1998; 11: 539–548. <u>https://doi.org/ 10.1023/A:1022315413516</u>
- Yun JH, Roh SW, Whon TW, Jung MJ, Kim MS, Park DS, et al. Insect gut bacterial diversity determined by environmental habitat, diet, developmental stage, and phylogeny of host. Appl Environ Microbiol. 2014; 80: 5254–5264. https://doi.org/10.1128/AEM.01226-14 PMID: 24928884
- Park R, Dzialo MC, Spaepen S, Nsabimana D, Gielens K, Devriese H, et al. Microbial communities of the house fly *Musca domestica* vary with geographical location and habitat. Microbiome. 2019; 7: 1–12. https://doi.org/10.1186/s40168-019-0748-9 PMID: 31699144

- Chandler JA, Morgan Lang J, Bhatnagar S, Eisen JA, Kopp A. Bacterial Communities of Diverse Drosophila Species: Ecological Context of a Host–Microbe Model System. Malik HS, editor. PLoS Genet. 2011; 7: e1002272. https://doi.org/10.1371/journal.pgen.1002272 PMID: 21966276
- Tougeron K, Tena A. Hyperparasitoids as new targets in biological control in a global change context. Biol Control. 2019; 130: 164–171. https://doi.org/10.1016/j.biocontrol.2018.09.003
- Cusumano A, Harvey JA, Bourne ME, Poelman EH, de Boer JG. Exploiting chemical ecology to manage hyperparasitoids in biological control of arthropod pests. Pest Manag Sci. 2020; 76: 432–443. https://doi.org/10.1002/ps.5679 PMID: 31713945
- Poelman EH, Harvey JA, van Loon JJA, Vet LEM, Dicke M. Variation in herbivore-induced plant volatiles corresponds with spatial heterogeneity in the level of parasitoid competition and parasitoid exposure to hyperparasitism. Funct Ecol. 2013; 27: 1107–1116. https://doi.org/10.1111/1365-2435.12114
- 69. Tiwari S, Rajak S, Mondal DP, Biswas D. Sodium hypochlorite is more effective than 70% ethanol against biofilms of clinical isolates of *Staphylococcus aureus*. Am J Infect Control. 2018; 46: e37–e42. https://doi.org/10.1016/j.ajic.2017.12.015 PMID: 29398071
- Kreve S, Reis ACD. Bacterial adhesion to biomaterials: What regulates this attachment? A review. Jpn Dent Sci Rev. 2021; 57: 85–96. https://doi.org/10.1016/j.jdsr.2021.05.003 PMID: 34188729
- Lommen A. MetAlign: Interface-driven, versatile metabolomics tool for hyphenated full-scan mass spectrometry data preprocessing. Anal Chem. 2009; 81: 3079–3086. <u>https://doi.org/10.1021/ac900036d</u> PMID: 19301908
- 72. Tikunov YM, Laptenok S, Hall RD, Bovy A, de Vos RCH. MSClust: a tool for unsupervised mass spectra extraction of chromatography-mass spectrometry ion-wise aligned data. Metabolomics. 2012; 8: 714–718. https://doi.org/10.1007/s11306-011-0368-2 PMID: 22833709
- 73. Dinno A. dunn.test: Dunn's test of multiple comparisons using rank sums. 2017. p. R package version 1.3.5. Available: https://cran.r-project.org/package=dunn.test
- R Core Team. A language and environment for statistical computing. Vienna, Austria: R foundation for statistical computing; 2022.
- 75. Caporaso JG, Lauber CL, Walters WA, Berg-Lyons D, Lozupone CA, Turnbaugh PJ, et al. Global patterns of 16S rRNA diversity at a depth of millions of sequences per sample. Proc Natl Acad Sci. 2011; 108: 4516–4522. https://doi.org/10.1073/pnas.1000080107 PMID: 20534432
- 76. Kozich JJ, Westcott SL, Baxter NT, Highlander SK, Schloss PD. Development of a dual-index sequencing strategy and curation pipeline for analyzing amplicon sequence data on the MiSeq illumina sequencing platform. Appl Environ Microbiol. 2013; 79: 5112–5120. <u>https://doi.org/10.1128/AEM.01043-13</u> PMID: 23793624
- Edgar RC. UNOISE2: improved error-correction for Illumina 16S and ITS amplicon sequencing. bioRxiv [preprint]. 2016; 1–21. https://doi.org/10.1101/081257
- Callahan BJ, McMurdie PJ, Holmes SP. Exact sequence variants should replace operational taxonomic units in marker-gene data analysis. ISME J. 2017; 11: 2639–2643. <u>https://doi.org/10.1038/ismej.2017.</u> 119 PMID: 28731476
- McKnight DT, Huerlimann R, Bower DS, Schwarzkopf L, Alford RA, Zenger KR. microDecon: A highly accurate read-subtraction tool for the post-sequencing removal of contamination in metabarcoding studies. Environ DNA. 2019; 1: 14–25. https://doi.org/10.1002/edn3.11
- Davis NM, Proctor DM, Holmes SP, Relman DA, Callahan BJ. Simple statistical identification and removal of contaminant sequences in marker-gene and metagenomics data. Microbiome. 2018; 6: 1– 14. https://doi.org/10.1186/s40168-018-0605-2 PMID: 30558668
- McMurdie PJ, Holmes S. phyloseq: An R package for reproducible interactive analysis and graphics of microbiome census data. Watson M, editor. PLoS One. 2013; 8: e61217. <u>https://doi.org/10.1371/</u> journal.pone.0061217 PMID: 23630581
- Oksanen J, Blachet F, Kindt R, Legendre P, Minchin P, O'Hara R, et al. Vegan: community ecology package. R Package vegan; 2015.
- **83.** Anderson MJ. A new method for non-parametric multivariate analysis of variance. Austral Ecol. 2001; 26: 32–46. https://doi.org/10.1111/j.1442-9993.2001.01070.pp.x
- Wang Y, Sun W, Fleischmann S, Millar JG, Ruther J, Verhulst EC. Silencing Doublesex expression triggers three-level pheromonal feminization in *Nasonia vitripennis* males. Proc R Soc B Biol Sci. 2022; 289: 1–10. https://doi.org/10.1098/rspb.2021.2002 PMID: 35078369
- Poelman EH, Gols R, Gumovsky A V., Cortesero AMA-M, Dicke M, Harvey JA. Food plant and herbivore host species affect the outcome of intrinsic competition among parasitoid larvae. Ecol Entomol. 2014; 39: 693–702. https://doi.org/10.1111/een.12150

- Cusumano A, Zhu F, Volkoff AN, Verbaarschot P, Bloem J, Vogel H, et al. Parasitic wasp-associated symbiont affects plant-mediated species interactions between herbivores. Ecol Lett. 2018; 21: 957– 967. https://doi.org/10.1111/ele.12952 PMID: 29656523
- **87.** Strand MR, Burke GR. Polydnavirus-wasp associations: Evolution, genome organization, and function. Curr Opin Virol. 2013; 3: 587–594. https://doi.org/10.1016/j.coviro.2013.06.004 PMID: 23816391
- Müller MJ, Dörr NCD, Deprá M, Schmitz HJ, Valiati VH, Valente VL da S. Reevaluating the infection status by the *Wolbachia* endosymbiont in *Drosophila* Neotropical species from the *willistoni* subgroup. Infect Genet Evol. 2013; 19: 232–239. https://doi.org/10.1016/j.meegid.2013.07.022 PMID: 23906981